

Barriers and enablers for the adoption of sustainable design practices using new design methods – Accelerating the sustainability transformation in the

Downloaded from: https://research.chalmers.se, 2024-10-20 01:23 UTC

Citation for the original published paper (version of record):

Mallalieu, A., Isaksson Hallstedt, S., Isaksson, O. et al (2024). Barriers and enablers for the adoption of sustainable design practices using new design methods – Accelerating the sustainability transformation in the manufacturing industry. Sustainable Production and Consumption, 51: 137-158. http://dx.doi.org/10.1016/j.spc.2024.08.023

N.B. When citing this work, cite the original published paper.

research.chalmers.se offers the possibility of retrieving research publications produced at Chalmers University of Technology. It covers all kind of research output: articles, dissertations, conference papers, reports etc. since 2004. research.chalmers.se is administrated and maintained by Chalmers Library

Contents lists available at ScienceDirect



Sustainable Production and Consumption



journal homepage: www.elsevier.com/locate/spc

Barriers and enablers for the adoption of sustainable design practices using new design methods – Accelerating the sustainability transformation in the manufacturing industry \star

Adam Mallalieu^{a,*}, Sophie Isaksson Hallstedt^{a,b}, Ola Isaksson^a, Matilda Watz^b, Lars Almefelt^a

^a Chalmers University of Technology, Division of Product Development, Department of Industrial and Materials Science, Hörsalsvägen 7A, 412 96 Gothenburg, Sweden ^b Blekinge Institute of Technology, Department of Strategic Sustainable Development, 371 79 Karlskrona, Sweden

ARTICLE INFO

Editor: Prof. Suiran Yu

Keywords: Sustainable design practices Sustainability transformation Industrial adoption Design methods Glaserian grounded theory

ABSTRACT

Product development and manufacturing organizations struggle in their sustainability transformation and do not sufficiently contribute to sustainable production and consumption. Design researchers, at the same time, develop and propose a plethora of new and improved design methods that can support the manufacturing industry in such transformation. It is, despite this, well-documented in literature that the industrial adoption of such proposed design methods is challenging. Previous research in the design domain has mainly studied this issue from a process and methodological perspective, whereas previous research in the management domain instead has focused on organizational, and human-behavioral aspects. This poses a research gap for more interdisciplinary research that studies the adoption of design methods from all three perspectives (i.e., process and methodology, organization, and human behavior). Six parallel case studies were carried out with three different product development and manufacturing organizations to collect qualitative empirical data. Glaserian grounded theory was used to analyze the collected data. This resulted in a descriptive framework that captures 53 interdisciplinary factors influencing the adoption of sustainable design practices using new and improved design methods. The descriptive framework is compared to interdisciplinary literature to further clarify and explain the findings, highlighting both practical and theoretical implications. This research provides three main contributions to theory and practice: (1) Two new concepts are introduced and used to explain the empirical findings, which are referred to as the dualism of design methods, and the situational design problem; (2) Nine systemic barriers and eight propositions are formulated, which highlight the need for a paradigm shift in how design is practiced in industry, how cognitive biases inside organizations can lead to a state of pseudo-sustainability, and the need for improved information and data management capabilities in organizations; (3) Sustainable design thinking is proposed as an potential enabler to address several of the main barriers, as it aims to provide a base competence of sustainable design to systematically challenge cognitive biases inside organizations.

1. Introduction

There is a growing attention and consensus regarding the need for a so-called *sustainability transformation* in the product development and manufacturing industry, i.e., new product development efforts that contribute to sustainable production and consumption (see e.g., Bengtsson et al., 2018). *Design*, or *design practices* (see e.g., Cross, 1992;

Dorst and Cross, 2001; Dorst, 2011), have both been highlighted to contribute to and strengthen organizations' innovation capabilities (see e.g., Micheli et al., 2018; Verganti et al., 2021), and critical in the *sustainability transformation* (Klotz et al., 2018; Sumter et al., 2020; Baldassarre et al., 2020), i.e., *sustainable design*, or *sustainable design practices*. Furthermore, previous research has shown that the appropriate adoption¹ of 'new and improved', or 'evidence-based', design

* Corresponding author.

https://doi.org/10.1016/j.spc.2024.08.023

Received 14 March 2024; Received in revised form 23 August 2024; Accepted 25 August 2024 Available online 28 August 2024

2352-5509/© 2024 The Authors. Published by Elsevier Ltd on behalf of Institution of Chemical Engineers. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

^{*} Research funded by Vinnova Project ID: 2020-04163.

E-mail addresses: adammal@chalmers.se (A. Mallalieu), sophie.hallstedt@chalmers.se (S. Isaksson Hallstedt), ola.isaksson@chalmers.se (O. Isaksson), matilda. watz@bth.se (M. Watz), lars.almefelst@chalmers.se (L. Almefelt).

¹ Adoption is here defined as "accepting or starting to use something new" (Cambridge, 2024). Adoption can however be partial, meaning that instances or facets of a design method are adopted.

methods proposed by design researchers can support product development and manufacturing organizations to develop such capabilities and transform their current design practices into sustainable design practices (Faludi et al., 2020; Hallstedt et al., 2023a). Such organizations are, despite the pressing need, not able to appropriately adopt such proposed design methods (Karlsson and Luttropp, 2006; Baldassarre et al., 2020; Faludi et al., 2020) and struggle to integrate social, economic, and ecological perspectives in their current design practices (Vilochani et al., 2024). Moreover, the early design phases have been highlighted as critical where the ability to anticipate sustainability performance (i.e., both social, ecological, and economical) and design products with a lifecycle perspective is considered crucial (Bhander et al., 2003; Ramani et al., 2010; Hallstedt et al., 2023b). A research project, spanning over two years, was therefore initiated with several Swedish product development and manufacturing organizations, which focused on demonstrating these organizations' ability to adopt sustainable design practices using 'new and improved' design methods proposed by design researchers. Six parallel case studies were carried out in the early design phases across these organizations to collect empirical qualitative data, using participant observation (Säfsten and Gustavsson, 2020).

The industrial adoption of 'new and improved' design methods proposed by design researchers has historically been considered low (see e.g., Araujo et al., 1996; Wallace, 2011; Gericke et al., 2020). Gericke et al. (2020) claim that design methods are used in conjunction with other design methods, and introduce the concept of method ecosystem, which refers to the unique process and methodological context of an organization, and argue that proposed design methods must be adapted and fit into this context. A recent study (Parolin et al., 2024) focuses on sustainability assessments in the early phases of design and they provide several "design propositions" on how sustainability assessments can be carried out in these phases, such as using qualitative indicators. Similar to the literature listed above, much of previous research focuses on the industrial adoption of design methods from a process and methodological perspective, and fewer contributions can be found that provide insights regarding human-behavioral and organizational aspects (López-Mesa and Bylund, 2011; Booker, 2012; Pieroni et al., 2019). Such literature is what we in this paper broadly categorize and refer to as literature in the design domain, where a 'pragmatic approach' is common and results in proposals on how design methods can be modified, or 'improved', to meet the needs of practitioners and fit an organization's current design practices, or method ecosystem. Furthermore, what we instead broadly categorize and refer to as literature in the management domain has, on the other hand, had more focus on organizational and human-behavioral aspects. There is a research gap and need for interdisciplinary research that explores and bridges these domains and frames the industrial adoption of design methods from all three perspectives. The following research question was thus formulated to capture all three perspectives during the empirical data collection:

• What influences the adoption of sustainable design practices using new and improved design methods?

Glaserian grounded theory (Walker and Myrick, 2006; Charmaz, 1996) was used to analyze the data and resulted in a descriptive framework that frames 53 interdisciplinary factors. The framework is also compared to interdisciplinary literature to further clarify and explain the empirical findings. The main contribution of this study serves as the base for new practical and theoretical insights towards:

- (i) How to facilitate and increase the adoption of new and improved design methods.
- (ii) How new and improved design methods can support practitioners in their organization's *sustainability transformation*.

The remainder of the paper is structured as follows: Section 2 provides the theoretical background within the studied topic; Section 3 presents the research methodology focusing on the research context, and how qualitative data was captured and analyzed; Section 4 presents the results from the analysis as a descriptive framework along with two new concepts; Section 5 provides a comparison between the empirical findings and literature to highlight the theoretical contributions and the main implications on practice; Section 6 provides conclusions and key points to research and practice, along with future work.

2. Literature review

This paper adopts an interdisciplinary focus, and Section 2 aims to provide readers from different disciplines with sufficient literature and background of all three perspectives (i.e., process and methodology, organization, and human behavior) to follow the reasoning throughout the paper. Section 2.1 provides theoretical background to design and designing, and the focus of this study. Section 2.2 provides the utilized lens for sustainable design practices. Section 2.3 presents a theoretical background to design methods and why they are relevant to practitioners. Section 2.4 provides a set of theories later used to explain the empirical findings, enabling us to bridge the identified research gap.

2.1. Design and designing

There have been many claims of what *design*, or *designing*, is (see e.g., Asimow, 1962; Jones, 1970; Schön, 1992). Simon (1969) stated that "everyone designs who devises courses of action aimed at changing existing situations into preferred ones" and that designing "is concerned with how things ought to be, with devising artifacts to attain goals". We view designing or any design process as a knowledge-producing process, where different design activities support designers to better understand the problem at hand as well as the potential solutions to that problem, and the overall goal is to identify a good problem-solution fit (Cross, 1992; Dorst and Cross, 2001; Dorst, 2006). Dorst and Cross (2001) refer to this as the *co-evolution of problem-solution*, which clarifies the iterative nature of design where both the problem and solution spaces evolve as we learn more about the problem and its potential solutions.

Eder (1998) divides a design process into the four generic phases of (i) understanding the problem, (ii) conceptualizing, (iii) embodiment, and (iv) detailing, and refers to (i) and (ii) as the "early stages of designing", which is the focus of this study. Here, design freedom is high, but less is known, and uncertainty is high, which results in a well-known dilemma referred to as the *design process paradox* (Ullman, 1992; Bhander et al., 2003; Chebaeva et al., 2021). *Front-loading* is a common strategy to counter the *design process paradox*, where the goal is to push knowledge generation as early as possible in the design process (Thomke and Fujimoto, 2000) and become critical to develop and propose more sustainable solutions (Bhander et al., 2003; Ramani et al., 2010; Hall-stedt et al., 2023b).

A design process is typically needed to treat design problems, i.e., illstructured, ill-defined, wicked, and unique problems (Simon, 1969; Rittel and Webber, 1973; Archer, 1979; Buchanan, 1992; Dorst, 2006; Gericke et al., 2022). Design problems are for example characterized as not having a single solution, unknown before the design process, and unique (Rittel and Webber, 1973; Buchanan, 1992). There have also been claims that design problems are paradoxical in the sense that designers do not know the 'real' design problem, or at least do not fully understand it before they have engaged in the early phases of design (Archer, 1979; Dorst, 2006). Potential solutions (i.e., 'products', 'designs', or 'artifacts') to such design problems are not obvious and benefit from applying designerly ways of knowing (Cross, 1982). Design problems can, for example, relate to 'how to transport users from A to B', 'how to design a lightweight airplane foil', 'how to design a circular supply chain', or 'how to design a music festival'. The paradigm of design thinking (Brown, 2008) has, for example, effectively utilized these insights of design or designerly ways of knowing, thinking, and acting (Cross, 1982; Cross, 2023) to revolutionize problem-solving activities on a

broader level outside of its origin within architecture and product design (Verganti et al., 2021; Auernhammer and Roth, 2021). To summarize, this sub-section underpins why *design*, or *design practices*, can be effective when solving complex problems related to sustainable development.

2.2. Sustainable design practices

Baldassarre et al. (2020) broadly define sustainable design as "the rational and structured process to create something new for solving sustainability-related problems" and argue that a complete transformation to *sustainable design practices* is a matter for the whole organization. Sala et al. (2015) highlighted the challenge of understanding "what contributes to a sustainable development and what does not". Our study utilizes the *Framework for Strategic Sustainable Development* (FSSD) (see e.g., Broman and Robèrt, 2017) to frame how design, or *sustainable design*, can contribute to sustainable development. Included in the FSSD are for example eight principles that define socio-ecological sustainability that need to be considered throughout a design's, or product's, full lifecycle to contribute to sustainable development. These eight principles are explicitly defined (Broman and Robèrt, 2017): "In a sustainable society, nature is not subject to systematically increasing…

- 1. Concentrations of substances extracted from the Earth's crust.
- 2. Concentrations of substances produced by society.
- 3. Degradation by physical means.

And people are not subject to structural obstacles to

- 4. Health.
- 5. Influence.
- 6. Competence.
- 7. Impartiality.
- 8. Meaning-making."

The FSSD also provides guidelines on how to adopt a strategic approach towards sustainable development. This includes the use of a forecasting and backcasting approach: (i) The vision is identified and formulated; (ii) The barriers to realizing the vision are identified and formulated; (iii) The required steps to reach the vision are identified and formulated; (iv) The different steps are prioritized. To summarize, we outline the following definition of *sustainable design practices* in this paper:

The appropriate incorporation of the *Framework for Strategic Sustainable Development* and *designerly ways of knowing, thinking, and acting* in an organization's design process.

Adopting *sustainable design practices*, in turn, strives to ensure that the developed solutions comply with the eight socio-ecological principles of the FSSD throughout their full lifecycle, and across the full value chain. This is, however, challenging and requires appropriate skills, knowledge, and the adoption of several design methods (Faludi et al., 2020; Hallstedt et al., 2023a).

2.3. Design methods and their role to designers

Design methods are one of the main study objects in our study and a term both referred to and used differently across research domains (see e.g., Bunge, 1966; Layton Jr, 1974; Niiniluoto, 1993; Araujo, 2001; Van Aken, 2004; Gericke et al., 2022). Cross (2000) broadly claimed that "in a sense, any identifiable way of working, within the context of designing, can be considered to be a design method". Jones (1970) argued that design methods are "attempts to make public the hitherto private thinking of designers; to externalize the design process" and Eder (1998) also describes them as "prescriptive knowledge as advice about designing ('know-how')". Wallace (2011) defines a design method as: "a prescriptive plan of action by which a class of design tasks are tackled".

Moreover, Daalhuizen and Cash (2021) stated that a design method is an encapsulation of procedural knowledge key to designing and the design process. Gericke et al. (2022) provide further nuance to what design methods are:

"A specification of how a specified result is to be achieved. This may include specifications of how information is to be shown, what information is to be used as input to the method, what tools are to be used, what actions are to be performed and how, and how a task should be decomposed and how actions should be sequenced".

To summarize, design method(s) as an umbrella term can thus generally be considered wide in what it covers, where different design methods treat different aspects of design, such as sustainability performance. They also apply to different situations, such as early phases, and types of products, or artifacts. We argue that the role of any design method is to prescriptively guide designers in generating a set of specific knowledge related to the '*design problem* at hand' (e.g., 'how to transport the user from A to B', or 'how to design a lightweight airplane foil'). In turn, design methods support designers in meeting the overall goal of designing, which is to identify a good problem-solution fit (Cross, 1992; Dorst and Cross, 2001; Dorst, 2006).

2.4. Adoption of design methods and perspectives on managing change

Design as a research domain, see e.g., Blessing and Chakrabarti (2009) and Gericke et al. (2022), proposes a plethora of, rigorously verified, new, and improved prescriptive design methods, and are occasionally also referred to as 'formalized', 'theory-based' and/or 'industry best' practices (Eder, 2009). Such design methods are typically developed and proposed to the industry by a design researcher where the *Design Research Methodology* is commonly used (Blessing and Chakrabarti, 2009). The adoption of such new and improved design methods proposed by design researchers has, however, had a low industrial uptake (Karlsson and Luttropp, 2006; Booker, 2012; Gericke et al., 2020; Faludi et al., 2020). The main reasons for the low uptake or adoption are summarized by Wallace (2011) explicitly:

- (i) Methods tend to be too complex, abstract, and theoretical.
- (ii) Too much effort is needed to implement them.
- (iii) The immediate benefit is not perceived.
- (iv) Methods do not fit the needs of designers and their working practices.
- (v) Little or no training and support are provided by companies.

Wallace (2011) does however also point out that designers do use design methods but often implicitly. Eder (1998) emphasizes the need for adapting proposed design methods and adds that practitioners need to claim ownership of design methods and refer to individuals who champion the design method inside an organization. Eder (2009) later refers to the concept of a method expert, which is an individual who has enough experience, competence, and understanding of the design method to ensure it is used as intended, and the term will later be used when referring to our study objects. Previous research provides several relevant insights concerning the challenges of adopting new and improved design methods proposed by design researchers in industry. However, adopting new and improved design methods requires further understanding and consideration of organizational, and humanbehavioral aspects (López-Mesa and Bylund, 2011; Booker, 2012; Pieroni et al., 2019). A set of concepts related to these perspectives are therefore briefly presented below and are later used in Section 5 to further highlight the need for considering all three perspectives, i.e., process and methodology, organization, and human behavior when studying the adoption of sustainable design practices using new and improved design methods.

Hackman and Oldham (1976) studied human-behavioral aspects concerning motivation and acceptance of new practices. They identified

autonomy as a critical component to 'work design' and defined autonomy as "the degree to which the job provides substantial freedom, independence, and discretion to the individual in scheduling the work and in determining the procedures to be used in carrying it out". The role of autonomy has been studied further, and Deci et al. (2017) argue in their Self-Determination Theory that basic psychological needs in work design are autonomy, competence, and relatedness. Furthermore, Kotter (1995) proposes one influential, but occasionally criticized (see e.g., Appelbaum et al., 2012), model on organizational change. Several recommendations are provided where, for example, the importance of involving people in the change process is emphasized. Kotter (2012) elaborates on Kotter's eight-step model and argues that the initial reason why change fails is due to complacency. Samuelson and Zeckhauser (1988) also emphasize that change is challenging in their status quo bias in decision making and that it is both easier, and common, to "doing nothing or maintaining one's current or previous decision".

Agency theory (see e.g., Ross, 1973; Mitnick, 1973; Eisenhardt, 1989) has been widely studied but in different research streams using different schools of thought and generally accounts for action in the relation between actors (i.e., principal-agent) and how it, for example, is influenced by e.g., bounded rationality, self-interests, and/or conflicting goals (Eisenhardt, 1989; Mitnick, 1992; Shapiro, 2005). The concept of *Bounded rationality* (Simon, 1969) can briefly be described as humans being biased and unable to make optimal decisions but rather making decisions that are satisfactory due to cognitive barriers i.e., satisficing. As a result, humans tend to use *heuristics*, or 'rules of thumb' in design, or 'complex problem-solving activities' (Simon, 1979). Simon (1969) elaborates on heuristics:

"Heuristic methods provide an especially powerful problem-solving and decision-making tool for humans who are unassisted by any computer other than their own minds, hence must make radical simplifications to find even approximate solutions".

In our study, it is important to acknowledge that some accounts treated in agency theory share similar traits to what others have referred to as cognitive barriers or *cognitive biases*, and are of relevance (Mitnick, 1992). Mitnick (2019) also explicitly stated, in a non-peer-reviewed paper, that "people make decisions based on things like norms, information with social origins, and what more recent literature terms cognitive heuristics or biases". The presence of norms or informal rules in organizations is further treated in works related to *Institutional theory* (see e.g., Meyer and Rowan, 1977; DiMaggio and Powell, 1983). DiMaggio and Powell (1983) explain institutions and introduce the concepts of *institutional rules* and *institutional isomorphic change* which govern how organizations behave, or act, and change according to norms and informal rules. To summarize, several theories can explain the underlying reasons for several barriers and enablers of the studied topic and will be further revisited in Section 5.

3. Research methodology

This research encompasses a large empirical study utilizing a multiple case study approach (Säfsten and Gustavsson, 2020), and the overarching research methodology is visualized in Fig. 1. This figure highlights the different activities (grey boxes), in- and outputs, and approximate order. Six industrial case studies (denoted as A1, A2, B1, C1, C2, and C3) were carried out during a two-year research project together with three Swedish product development and manufacturing organizations. Each of the case studies had a different sustainable design scope, further described in Section 3.2. The research project as such focused on demonstrating these organizations' ability to adopt *sustainable design practices* using new and improved design methods proposed by design researchers. This allowed us to the socio-technical interplay between design researchers, practitioners, and design methods using participant observation (Säfsten and Gustavsson, 2020). This is further described in both Sections 3.2 and 3.3. In turn, this resulted in the



Fig. 1. Applied research methodology highlighting the analysis procedure, supplementary activities, and in/outputs.

collection of a large set of qualitative data (Säfsten and Gustavsson, 2020) relevant to the scope and the formulated research question in this paper. As visualized in Fig. 1, different types of qualitative data were collected from the six case studies and are further described in Section 3.3. The data was collected using different activities in the multiple case studies and is further described in Section 3.2. Glaserian grounded theory (Walker and Myrick, 2006) was identified and chosen as an appropriate method for analyzing the collected qualitative data, and developing a descriptive framework. Charmaz (1996) describes Grounded theory as a set of inductive strategies for analyzing data, which can be used to synthesize, explain, and understand the collected data. It builds on both interpretative as well as positivist assumptions and has a strong history in qualitative research, which allows both different and novel themes to emerge. It can also support making conceptual sense of large sets of qualitative data. As visualized in the surrounding blue box in Fig. 1, the analysis procedure and development of a descriptive framework using Glaserian Grounded Theory followed three main steps: (1) Open coding; (2) Selective coding; (3) Theoretical coding (Walker and Myrick, 2006). The first coding step captured several different aspects related to the adoption of sustainable design practices from one or more of the three perspectives (i.e., process and methodology, organization, and human behavior). The latter two coding steps further nuanced these aspects, which resulted in a final descriptive framework consisting of 53 interdisciplinary factors that influence the adoption of sustainable design practices using new and improved design methods. The analysis procedure is further described in Section 3.5 and provides further clarity on how the codes as such relate to the adoption of sustainable design practices. Additionally, the developed framework was lastly compared with the literature provided in Section 2 to further nuance and explain the findings, which is also visualized in Fig. 1. Altogether, the research methodology supports fulfilling the explorative and interdisciplinary scope of our study, making it a suitable choice. The sub-sections below are dedicated to describing the analysis procedure and other relevant aspects related to the research methodology in more

detail. This includes the utilized design methods, the industrial cases, the data collection, and the limitations of the study.

3.1. Design methods used in the multiple case study

The design methods used in the multiple case studies are divided into two categories in this paper.

Category one: Systems Engineering Design (SED) design methods, based on more than 20 years of research in close collaboration with the Swedish product development and manufacturing industry (see e.g., Isaksson et al., 2000; Isaksson et al., 2013; Borgue et al., 2021). The SED design methods aim to support the modeling and design of complex systems or designs where there are several dependencies and interactions between different domains and sub-systems (e.g., functions in the system, stakeholders in the system, and/or engineering disciplines utilized in the system). These dependencies are evaluated jointly in the early phases of design, which ultimately supports practitioners in assessing the impacts of different design decisions on complex sociotechnical systems (Isaksson et al., 2023). Applying the SED design methods typically involves the collection of data and information from different stakeholders in the organization which is later incorporated and/or modeled in computer-based tools by either a design researcher or practitioner, i.e., a method expert. The outcome is later communicated and utilized across organizations to enable more information-based design decisions.

Category two: Sustainable Product Development (SPD) design methods and are based on more than 15 years of research in close collaboration with the Swedish product development and manufacturing industry (see e.g., Byggeth and Hochschorner, 2006; Hallstedt et al., 2013; Watz and Hallstedt, 2022). The SPD design methods aim to support the strategic integration of socio-ecological sustainability, using the FSSD (Broman and Robert, 2017), in product development and manufacturing organizations by ensuring that a full systems perspective is incorporated in the early phases of design. The design methods support practitioners in e.g., anticipating the sustainability performance of different solutions, identifying sustainabilityrelated risks, and guidance towards more sustainable solutions. Applying the SPD design methods typically involves a 2–3 h facilitated workshop using a multidisciplinary team of practitioners, including different designers, experts, and/or specialists inside the organization, where a set of key questions or focused topics are discussed. This information is later consolidated either by design researchers or practitioners, i.e., a *method expert*, and the outcome is typically communicated and used by relevant stakeholders inside and outside the organization.

3.2. Case companies and activities in the multiple case study

Three different Swedish companies in three different sectors were used in the multiple case studies. The case companies are suitable choices as they represent: (i) International product development and manufacturing organizations; and (ii) Organizations that have an expressed ambition to design and manufacture more sustainable solutions, and thus transform their current design practices into *sustainable design practices*.

Company A is a large first-tier supplier that develops and manufactures integrated metallic and composite assemblies for aerostructures and aero-engine products. Two case studies were carried out with Company A, both focusing on the sustainable industrialization of additive manufacturing. The cases were with different scopes and performed with different functions within the organization but both within the context of technology integration and development.

Company B is a semi-large product development and manufacturer of sealing solutions for the telecom, manufacturing, and construction industries. This case focused on sustainability governance and how to integrate sustainability in their product innovation process, where a steel frame was used as the case product. **Company C** is a large product development and manufacturing company in the automotive sector. Three different cases were carried out with this company, but two had several joint activities, namely cases C1 and C2 (see Table 1). The scope of these was deemed to be aligned and required tighter collaboration between the two business functions (product design, and procurement and sourcing). This required participants from several different functions since product design and procurement and sourcing are treated as two different business units within the same company. The third case from this company came from a third business unit and was treated as a separate case study.

The case studies represent six different design scopes and five different organizational functions, as illustrated in Table 1. The design methods applied in each case study were identified together with the industrial practitioners individually for each case, i.e., which design methods were deemed to suit the challenges addressed in the design scopes. Each of the six cases involved more than 10 different industrial participants and five different design researchers. Furthermore, as visualized in Fig. 1, the research project and its six parallel industrial case studies (A1–2, B1, and C1–3) served as input for collecting relevant qualitative data for two years, and in four distinct ways:

- (1) A total of 33 workshops were carried out across the six case studies and corresponded to the appropriate use of many of the proposed design methods, as described in Section 3.1.
- (2) Recurring bi-weekly meetings varying between 30 and 45 min were held with each company to coordinate the activities carried out in the case studies. These meetings were, for example, used to ensure that relevant industrial participants took part in the workshops and that appropriate data and information were at hand in the workshops or fed into the design methods. These also involved practitioners elaborating on issues they are facing regarding their sustainability transformation.
- (3) Four large seminars were carried out in the research project with the full project consortium that consisted of an additional four product development and manufacturing organizations, and two product development consultancy organizations. The seminars included several emerging topics concerning the research question and were discussed in larger workshop formats.
- (4) More than ten surveys were sent out either during or in preparation for the four seminars to obtain individual practitioner responses on the topics discussed during the seminars. Qualitative data was collected as practitioner responses in the shape of freeform text.

Both design researchers and practitioners were treated as study objects since they both are considered stakeholders relevant to the research question. Practitioners served as the practitioners or users of the design methods while also representing the organizations' challenges towards a *sustainability transformation*. The design researchers were divided into two roles, both as *method developers*, i.e., the ones that have developed the design method, but also as the role of *method experts*. It is not necessarily the design researchers who take the role of *method experts* but took this role in many instances since most of the design methods are new to the organizations. Moreover, it was thus also in most instances the *method developers* i.e., the design researchers, who proposed the design methods to the organizations according to their expressed challenges in this study.

3.3. Qualitative data collection using participatory observation

One researcher, who had no prior experience in using the proposed design methods, took part in 30 out of the 33 conducted workshops, and all four seminars. This researcher also acted as coordinator of the six parallel case studies and thus participated in most of the informal meetings. In turn, qualitative data was collected using *participant observation* (Säfsten and Gustavsson, 2020). Furthermore, the degree of

Table 1

Tabl	C I
Caro	docaription

Gase ac	scriptions.					
Case	Company	Context	Design scope	Conducted workshops	Nr. of different SED methods applied	Nr. of different SPD methods applied
A1	А	Technology integration and development	Sustainable design of a turbine rear structure using laser powder bed fusion	2 separate, and 2 joint with A2	1	3
A2	А	Technology integration and development	Sustainable repair of a fan blade with direct energy deposition	2 separate, and 2 joint with A1	-	3
B1	В	Management and product design	Strategically integrating sustainability in the product innovation process of a steel frame	8 separates	-	5
C1	С	Product design	The sustainable and circular design of a seat in the new generation of electric vehicles	10 joint with C2, and 1 joint with C2 and C3	3	6
C2	С	Procurement and sourcing	Sustainable and circular supply chains for a seat in the new generation of electric vehicles	1 separate, 10 joint with C1, 1 joint with C1 and C3	3	6
C3	С	Material and product design	Sustainable material selection for a cable bracket component	7 separate, and 1 joint with C1 and C2	1	3

participation was *moderate* where "the observer balances between being an insider and an outsider, between participation and observation" (Säfsten and Gustavsson, 2020). Qualitative data was collected during workshops, seminars, and informal meetings, and in most instances in line with what can be defined as *qualitative observations* (Creswell, 2014) and what Säfsten and Gustavsson (2020) refer to as *direct* and *unstructured* in the format of in-vivo (Miles et al., 2014) in notebooks. The collected qualitative data was, however, not only in the form of qualitative observations and was deemed to be distinguishable into four different types:

- (1) Quotes: either from a design researcher or a practitioner.
- (2) Observations: Concrete events or observations that occurred, either by a design researcher or practitioner.
- (3) Notes: Either something a design researcher or practitioner said or presented but not captured how it was explicitly phrased and are therefore separated from quotes. It also relates to assigned 'action points', either by a design researcher or practitioner.
- (4) Reactions: The observer's reaction to observations, which resulted in either a 'thought' or 'idea' related to the topic that is studied.

One example of each type of collected data is provided in Table 2.

Qualitative data was also collected using surveys via online survey tools. Data was collected for almost two years and resulted in almost five full notebooks, where each included 192 A5 pages of qualitative data. The collected data captured aspects relevant to the topic and included both process and methodological, organizational, and humanbehavioral aspects.

3.4. Limitations of the study

Three main limitations can be linked to this study: (1) A large amount of data was generated, a recurring dilemma in Glaserian Grounded Theory due to its explorative nature (Cutcliffe, 2000). The data used in steps one and two was therefore strictly limited to cases C1 and C2 and for a fixed period to avoid biased selection of data; (2) The

data was collected during specific occasions in a research project, as specified in Section 3.2, and not within the daily operations of the organizations and can influence potential claims. The appropriate use of the design methods was however studied in real cases as part of in-house projects in the organizations; (3) One individual researcher oversaw data collection and the Glaserian grounded theory analysis. A correspondence criterion was therefore used throughout the coding process, and steps one and two were also rigorously checked by two researchers to reduce the risk of bias. The overall procedure is further clarified below.

3.5. Analysis of empirical data using Glaserian grounded theory

As mentioned, Glaserian grounded theory was used to analyze the empirical data, as visualized in Fig. 1, and is separated into three main steps (Walker and Myrick, 2006): (1) Open coding; (2) Selective coding; (3) Theoretical coding. Moreover, three guiding principles were used to preserve the interpretative and positivist assumptions of Glaserian grounded theory (Åge, 2011):

- (1) The codes correspond to the data (positivism).
- (2) The data is interpreted and understood in its context when assigned codes (hermeneutics).
- (3) The codes are useful (pragmatism).

3.5.1. Open coding of case studies C1 and C2

Before the open coding could be carried all collected data from case studies C1 and C2 was inserted manually from the notebooks into Excel to ease the coding process. The open coding later began by assigning individual meaning, i.e., open codes, to each line of the collected data line by line and comparing them to each other. Approximately 220 lines of data captured in the Excel file were analyzed and resulted in roughly 200 open codes and two examples from the analysis are presented in Table 3.

In example one, see Table 3, row two, data was captured in an informal meeting with company C where a practitioner elaborated on

Table 2					
Evennlee	of collected	data	for	oooh	trme

Types of collected qualitative data	Example of how it was explicitly captured in the notebook	Context of where data was collected
Quote	"The participating practitioners want to know they will get 'something out' from taking part in the workshops"	This was a quote by one practitioner during an informal coordination meeting
Observation	Difficult to comprehend/limit the discussion related to the specific case product	This observation was captured during the use of one of the design methods focusing on assessing the sustainability performance of the case product
Note	Must check what data we already have that can be used, and then 'bounce it' with the company coordinator	This was a note related to planned activities to be carried out with one of the case companies
Reaction	Do the customers even want to have transparency? They might be shocked when they see the social issues in the value chain	This was a reaction noted down during a workshop focusing on assessing the sustainability performance of on case product.

Table 3

Examples from the open and selective coding procedure.

Context	Date	Note	Open coding	Selective coding
Informal meeting with company C - Procurement and sourcing	210614	Practitioner: *Need to work in new ecosystem to work "circular" *See a need of a new system *Various of different new issues "1000's" *Traceability, several actors -> need to ensure "waste" is managed -> How do we even start this collaboration	There is a need to collaborate with actors differently to develop more sustainable solutions	New sustainable design practices
WS at company C where they present the case product for C1 and C2	220310	Development of product -Product breakdown -> different view -Core structure *Comfort system *Safety system *Custom specific -Product one of the most complex parts *Legal *Safety *Integrate many things -Scalable	There is a need for design researchers to understand the case product	Method experts' understanding of company case and context

issues and challenges related to the design of circular supply chains. The open code ("there is a need to collaborate with actors differently to develop more sustainable solutions") was assigned by interpreting and understanding the data based on who was involved and the activity (Context and date), together with what happened and/or what was said. This is also in correspondence with what the original data stated and useful to the research question. In the second example, see Table 3, row four, data was captured during a workshop arranged at company C for practitioners and design researchers involved in C1 and C2. The workshop focused on presenting the product used in the case study, and the data related to the product itself. The open code ("there is a need for design researchers to understand the case product") was assigned based on interpreting and understanding the data. This followed the principles above and is both in correspondence to the original data and useful to the research question.

3.5.2. Selective coding of case studies C1 and C2

The selective coding focused on identifying core categories based on the already identified open codes, where each open code was assigned a selective code with a higher abstraction level. Charmaz (1996) refers to this as "a category may subsume common themes or patterns in several codes", and two examples of this are provided in Table 3, column 5. The selective coding was carried out after the open coding was completed, but patterns and themes started to emerge during the open coding. The first selective code, or core category, was generated based on the first line of open code, and every open code that fit within that core category was assigned to it. New core categories were identified and formulated whenever the current set of core categories had a poor fit to an open code. Furthermore, open codes can be assigned to several core categories, which in turn meant that whenever a new core category was identified it also iteratively had to be compared to previous open codes to check potential fit. Eight core categories were generated in total, and these are presented in Table 4.

3.5.3. Theoretical coding of collected data

Theoretical memos were continuously formulated throughout the open and selective coding process, as visualized in Fig. 1. Such memos refer to preliminary ideas, or sub-concepts, captured as short sentences, and saturate and converge over time (Walker and Myrick, 2006). The theoretical coding as such later focused on iteratively going back and forth between the raw data, open codes, core categories, and theoretical memos. This step conceptualizes and frames how the open codes and core categories relate to each other (Walker and Myrick, 2006). Furthermore, data captured from the remaining cases (A1–2, B1, and C1–3) and additional sources, such as the surveys, were also used as input to the theoretical coding. This is also visualized in Fig. 1. Charmaz (1996) refers to this step as theoretical sampling, i.e., "collecting more data to clarify your ideas and to plan how to fit them together". No additional core category was identified, but this instead generated more open codes related to the core categories, which in turn:

- (i) Further clarified the core categories and additional relations between core categories.
- (ii) Strengthened the core categories and the identified relations.
- (iii) Added further depth to the core categories and the relations.

The results from the theoretical coding are captured in a descriptive framework that frames 53 interdisciplinary factors, and a summarizing table, presented in Section 4. Additionally, the full analysis of the theoretical coding is provided as supplementary information.

Table 4

Core ca	tegories	generated	during	the	selective	coding
---------	----------	-----------	--------	-----	-----------	--------

	Core categories	Frequency of core category in the open codes
Α	Practitioners' understanding of why and how to use design methods	45
В	Method developers' understanding of practitioner needs	37
С	Design methods fit into the current design processes	44
D	New sustainable design practices	53
E	Method experts' understanding of company case and context	47
F	Practitioners' design method engagement	35
G	Design method synergy and integration	38
Н	Information and data capturing in sustainable design practices	37

Table 5

ine of metors captured by the descriptive framework.	
Factor	Occurrence and degree of influence (see also Fig. 2)
F1: Lack of understanding leads to misuse	It was frequently observed that a lack of understanding of 'why and how' to use design methods had an impact on design method outcomes. In turn, this either had a reduced and/or negative impact on the sustainability performance of a solution and became a barrier to the center circle.
F2: Lack of understanding limits practitioner engagement	It was frequently observed that a lack of understanding of 'why and how' to use design methods resulted in decreased engagement by the practitioners. This factor becomes a barrier towards the center circle.
F3: There are difficulties in understanding how sustainability can be simplified	It was frequently observed that practitioners require or explicitly ask for some pragmatism in the context of sustainable design practices. This was typically addressed by simplifications or defining what they deemed as reasonable boundary conditions, i.e., narrowing the scope. It was, however, occasionally difficult to understand to what extent the scope could be narrowed, which led to design method outcomes that either had a reduced and/or negative impact on the sustainability performance of a solution.
F4: The concept of lifecycles is difficult to understand	Sustainable design practices require products to be designed for and assessed throughout all their lifecycle phases, i.e., material extraction to end-of-life and/or circular activities to extend the product's life or components life (e.g., repurpose or remanufacture). This challenged how the practitioners work at present, and it was frequently observed as difficult to either fully comprehend and/or practice this concert. This in turn, becomes a barrier towards the center circle.
F5: Failing to meet needs limit adoption	Several of the identified factors relate to the different needs of the practitioners. Some needs were observed, but some needs were occasionally explicitly expressed needs by practitioners. Failing to meet either of these becomes a direct barrier towards the center circle.
F6: Need to understand how methods can be fitted into the current design processes	It was frequently observed that fitting design methods to companies' current design process needs to be considered when introducing new design methods to practitioners. It was in addition to this also frequently mentioned by practitioners as a core need. In turn, this becomes a barrier towards the center circle.
F7: Need to understand how practitioners currently work	It was frequently observed that there is a need to understand how practitioners currently work to ensure applicability. Failing to meet this need will limit adoption and can also result in instant rejection, becoming a barrier towards the center circle.
F8: Need to understand how to get practitioners' engaged	It was occasionally observed that it was not fully clear how to get practitioners engaged. Failing to engage practitioners can have consequences on different scales and is a barrier to the center circle. It is not fully clear how to effectively, and efficiently, convert the "why and how" to use design methods.
F9. Need to be able to convey the willy and now to the practitioners	this not fully clear how to encertively, and enclendly, convey the why and how to use design methods since practitioners frequently require support in understanding the "why and how". Failing to meet this need results in either limited adoption and/or misuse. Successfully meeting this need can instead be an enabler towards the center circle.
F10: Adaptation is required	This barrier was constantly observed but to different degrees in terms of how much adaptation was needed. It could for example be that guiding questions in templates needed to be changed, that the representation of input- information and data needed to change, or that the scope and core idea of the design method had to be changed. In turn, this becomes a barrier towards the center circle.
F11: Sustainable design is considered complex and needs to be simplified	It was frequently observed that practitioners feel the need to simplify sustainability since it is a complex and systemic topic with many effects that are difficult to anticipate. This resulted in design method outcomes that, in turn, either had a reduced and/or negative impact on the sustainability performance of a solution and thus become a barrier towards the center circle.
F12: Increases if fitted into the current design process	It was frequently observed that if a design method was perceived as potentially possible to fit into the current design process, practitioner engagement increased and thus became an enabler towards the center circle.
F13: Limited if not fitted into the current design process	Practitioners frequently highlight the importance of being able to fit the design methods into the current design process and that adoption will be limited otherwise.
F14: Design process differs from company to company	It was constantly observed that the design processes differ from company to company, such as daily routines and design activities, available time and resources, competence, and the background of the design method user. This makes it difficult to ensure that design methods fit every individual design process since it will require some adaptation and thus becomes a barrier towards the center circle.
F15: The methods need to support challenges or 'problems' related to sustainable design practices	It was constantly observed that the design methods do support challenges or 'problems' related to sustainable design practices, which incentivized the practitioners to use the design methods. In turn, this becomes an enabler towards the center circle.
F16: Radical changes limit the ability to fit into the current design process	It was occasionally observed that the new sustainable design practices require larger or radical changes in how practitioners work today. One example relates to the sustainability criteria that the practitioner at this moment needs to consider in their early design phases, which currently (mainly) considers product material, geometry, and components. New sustainable design practices will however require the full lifecycle to be considered and designed for in the early design phases, and potentially circular services i.e., activities in the up- and downstream supply chains. In turn, this requires new types of input- data and information used in the design activities in the current design process and increases the scope of the design activities. Practitioners were occasionally observed as having reservations when the design methods were perceived as introducing such changes. In turn, this becomes a barrier towards the center circle.
F17: Requires new working practices and skills	It was frequently observed that new sustainable design practices require new working practices and skills. It was constantly observed that practitioners were reserved when the design methods signaled that new working practices needed to be introduced, and the practitioners would let the method experts know that 'this is not how we work today'. Furthermore, it was occasionally observed that there was a lack of certain specialist expertise required to use the design methods. It could either be that the practitioner required was unavailable due to limited time and prioritization or that the company lacked that role. It was frequently observed that there was a knowledge gap related to sustainability. In turn, this becomes a barrier towards the center circle.
F18: Sustainable design practices require the 'right' competence	Many of the new sustainable development practices in early phase design are qualitative design methods, and it was frequently observed that they require the 'right' competence when using the design methods. Using the design methods without the 'right' competence resulted in design method (continued on next page)

Factor	Occurrence and degree of influence (see also Fig 2)
ration	occurrence and degree of influence (see also rig. 2)
	outcomes that either had a reduced and/or negative impact on the sustainability performance of a solution, and thus becomes a barrier towards the center circle.
F19: Sustainability principles are not understood	It was frequently observed that core principles such as the three sustainability dimensions of social.
	ecologic, and economic along with a full lifecycle perspective are not well understood nor practiced at
	this moment. In turn, this resulted in design method outcomes that either had a reduced and/or
	negative impact on the sustainability performance of a solution and thus became a barrier towards the
	center circle.
F20: Methods require practitioners' engagement	It was constantly observed that the design methods require the practitioners' engagement.
	Practitioners were either required to perform certain actions and provide input- data and information
	or provide understanding such that method experts could perform tasks related to the design methods
E21. There is some superspace that sustainable design practices are peeded	In turn, this becomes a barrier towards the center circle.
F21: There is some awareness that sustainable design practices are needed	It was occasionally observed that the practitioners are aware that a transformation and change of practices are needed to develop more sustainable solutions. This had a positive influence on the
	incentives to use the design methods and thus became an enabler towards the center circle
F22: There is limited sustainability relevant expertise	It was frequently observed that there is a lack of the sustainability-relevant expertise that is required in
	new sustainable design practices. There was expertise within niche sustainability topics such as
	environmental regulations, substances of concern, or corporate social responsibility processes.
	Sustainable design practices do however require other types of expertise as well such as cross-
	disciplinary expertise within both design and sustainability. In turn, a lack of this becomes a barrier to
	the center circle.
F23: Several methods are needed	It was constantly observed that the companies are facing several challenges or 'problems' related to
	sustainable design and that each design method only addresses a set of these challenges or 'problems'.
	Fitting several design methods adds complexity. In turn, this will become a barrier towards the center
F94. Different to be our which motheds are needed	CITCLE.
F24: Difficult to know which methods are needed	there are also several design methods developed to choose from. These design methods vary in score
	along with their applicability to case and context. It was frequently observed as difficult for method
	experts to instantly know which design methods to use for the specific challenge or 'problem'. This
	becomes a barrier towards the center circle when combined with other barriers, such as F7 ("need to
	understand how practitioner currently work") and F33 ("difficult to get full understanding").
F25: There is a need to train and educate practitioners within sustainable	It was frequently observed that the competence and knowledge within sustainability need to increase
design	which in turn becomes a barrier towards the center circle. However, it was also observed that
	initiatives and resources are being allocated by the companies to address this. This, in turn, allows
	practitioners to allocate time to test and use new design methods, and it was occasionally observed as
EQC. Custoinshility relevant information and data need to be contrared	incentivizing to practitioners which has positive effects on Category F.
F26: Sustainability-relevant information and data need to be captured	It was irequently observed that sustainability-relevant data need to be captured, either due to uncoming legislation and /or company specific needs related to corporate targets and visions
	Furthermore, this was observed to incentivize practitioners to use design methods when combined
	with barrier F50 ("do not know what or how to capture") and enabler F51("supports to convey what
	sustainability relevant information and data to capture"). In turn, this becomes an enabler to the center
	circle.
F27: Requires collaboration with new internal and external actors	It was frequently observed that new sustainable design practices require practitioners' to collaborate
	with new actors both internally (e.g., actors from procurement or production, and different specialists)
	and externally (e.g., new suppliers, and actors enabling circular activities). This will enforce changes in
E20. Mathed applicability limited with out up donaton ding	how practitioners currently work and thus become a Darrier towards the center circle.
F28: Method applicability limited without understanding	It was frequently observed that when a method expert facked understanding of company case and context the applicability of the method was low. Lack of understanding will either limit the ability to
	adapt the design method such that it becomes applicable, or worse, result in an inability to understand
	if the design method is relevant to their specific challenge or 'problem'. This becomes a barrier towards
	the center circle and is also amplified by barrier F33 ("difficult to get full understanding"). It does
	however become reduced by enabler F34 ("practitioners' engagement provides understanding") which
	supports method experts to adapt the design method and increase applicability.
F29: Eases if company case and context is understood	It was frequently observed that understanding of company case and context supported both
	understanding if the design method has the potential to fit, but also how to adapt the design method
	such that it fits the current design process. In turn, this becomes an enabler towards the center circle
F30: Understanding enables appropriate simplification	It was frequently observed that understanding of company case and context supported reducing the
	scope of the design methods (i.e., simplifying) such that the minimum accepted boundary conditions
	considered complex and needs to be simplified") which ensures that misuse is avoided and thus
	becomes an enabler towards the center circle
F31: Method experts' interfere when practitioner fails to understand	Some design methods were either applied in a workshop format with a method expert as a facilitator of
1 1	in a format without the support of a facilitator. In the first scenario, which was more common than the
	second, method experts frequently interfered with practitioners to clarify 'why and how' to use the
	design method. In the second scenario, method experts instead supported by assessing the design
	method outcomes and occasionally had to adjust the design method outcomes. This supported in
	reducing the risk of design method outcomes that had a reduced and/or negative impact on the
	sustainability performance of a solution and thus becomes an enabler towards the center circle.
F32: Limited understanding can make integration difficult	It was trequently observed that method experts' understanding of case and context is important when
	integrating several design methods. Understanding is crucial when adapting design methods such that
	the output of one design method is used as input to another similar to fitting design methods. In St required when
	current design processes. This does for example involve considerations of design method scope, the
	representation of information and data that is required. This does not act as a harrier towards the
	center circle, but as a barrier towards factors F38-F42 and reduces their enabling effects on the center
	circle.

(continued on next page)

Table 5 (continued)

Eactor	Occurrence and degree of influence (see also Fig. 2)
	occurrence and degree of innuence (see also Fig. 2)
F33: Difficult to get full understanding	It was frequently observed that it is difficult for method experts to get a full understanding of the company case and context. The companies studied are large organizations with hundreds of employees, that all have different roles, internal processes, daily routines, along with complex products. This makes it difficult to get a full understanding of all relevant information and knowledge needed to manage barriers such as F7 ("need to understand how practitioners currently work"), F10 ("adaptation is required"), along with enabler F30 ("understanding enables appropriate understanding").
F34: Practitioners' engagement provides understanding	It was frequently observed that practitioners provide direct input that supports method experts to better understand how to adapt design methods such that they become relevant for the specific challenge or 'problem' and applicable to the specific context. Examples of this are when practitioners described their challenges or 'problems' in detail and asked for desired design method outcomes. Furthermore, it was also frequently observed that practitioners provided method developers with input on the company case and context. Examples of this are drawings, functional descriptions, internal process flows, and available resources. This increased the method experts' understanding, such that they could adapt the design methods accordingly. In turn, this becomes an enabler towards the center
F35: Practitioners have limited time	It was constantly observed that practitioners had limited time available to use design methods. Furthermore, it was frequently expressed and observed that they have limited time. Almost every meeting started with a practitioner either excusing themselves for not having been able to put down the time required for a certain task and/or saying that 'it is stressful at the moment because there is so much going on in our company'. This is both a direct barrier towards the center circle but also amplifies several other barriers that, in turn, also act as barriers towards the center circle.
F36: Practitioners ask for new methods that supports them	It was frequently expressed by practitioners explicitly that they need new design methods that support them in challenges or 'problems' related to sustainable design practices. However, because of many other identified barriers, this factor alone is not enough to satisfy the center circle but becomes an indirect enabler towards the center circle.
F37: Practitioner engagement is limited to AS IS	It was constantly observed that practitioners engage in design methods with the mindset of 'AS IS'. There was a reluctance to change according to what the design method prescribed along with its outcomes. Practitioners constantly brought up the current design practices and clearly stated that design methods cannot affect how the practitioners' currently work. This resulted in design method outcomes that had either a reduced and/or negative impact on the sustainability performance of a solution.
F38: Methods address different sustainable design challenges or 'problems'	It was frequently observed that different design methods address different challenges or 'problems' related to sustainable design practices which incentivize practitioners to use different design methods for different challenges or 'problems'. In turn, this had an enabling effect on barrier F23 ("several design methods are needed") and thus acted as an enabler towards the center circle.
F39: Methods can have synergies with existing methods	It was occasionally observed that the new design methods were possible to integrate with design methods in the existing design processes while also resulting in synergies. There were no observations of how design method outcomes eventually were used in existing design methods due to the period of the study, but practitioner excitement was observed when potential synergies were identified. This can potentially act as an enabler towards the center circle.
F40: Can integrate sustainability in other methods	It was occasionally observed that the new design methods were possible to integrate with design methods in the existing design processes while also integrating sustainability into these. There were no observations of how sustainability eventually was integrated into existing design methods due to the period of the study, but the potential was occasionally observed. In turn, this can act as an enabler towards the center circle.
F41: Methods applied in conjunction can generate a larger knowledge base	It was frequently observed that applying design methods in conjunction i.e., using the design method outcome from one design method into another, supported in the knowledge generated by the design method outcome of that design method. In turn, this acts as an enabler towards the center circle.
F42: Synergies can increase the incentive to adopt	It was frequently observed that integrating design methods increased the incentive to adopt design methods. Having a prescribed logic of how design method outcomes can be built upon and integrated into another design method was explicitly desired by practitioners, as it helps to clarify when and how the design methods should be connected. In addition to this, combining design methods and creating synergies supported addressing a wider range of challenges or 'problems' related to sustainable design practices and was also claimed to be desired. In turn, this incentivized practitioners to adopt design methods
F43: Requires lot of coordination since many actors need to be involved	It was frequently observed that integrating design methods and using several in conjunction required coordination between several actors. This required detailed planning and coordination, which was time-consuming and non-straightforward. This is not a barrier towards the center circle but reduces the persible anabling effect of design method superside
F44: Limited when there is lack of appropriate data	It was frequently observed that the data and information required carrying out the design methods are lacking in the companies. This either reduced the applicability of the design method, led to assumptions, limitations, and/or reduced the scope of the design method. This results in design method outcomes that have a reduced or negative impact on the sustainability performance of a solution.
F45: The information captured needs to be understood	It was frequently observed that the information captured in sustainable design practices needs to be understood. It was occasionally observed that the information was not well understood. Furthermore, it was occasionally also explicitly stated by practitioners that sustainability-relevant information is understood differently both inside the company and by external factors such as suppliers. This can, in turn, result in reduced and/or negative design method outcomes that impact the sustainability performance of a solution.
F46: Information from preceding activities need to be captured	It was constantly observed that information generated from preceding activities in the current design process and/or new design methods need to be captured, such that it can be used as input in the succeeding design method. This is however influenced by barrier F50 ("do not know what or how to capture"). Thus, acts as a barrier towards design methods' fit into the current design process, which in turn becomes a barrier towards the center circle

(continued on next page)

Factor	Occurrence and degree of influence (see also Fig. 2)
F47: Information from preceding activities need to be captured	It was frequently reported that information generated from preceding activities in the current and/or new design methods need to be captured, such that it can be used as input in the succeeding design method. This is however influenced by barrier F50 ("do not know what or how to capture"). Thus, acts as a barrier towards integrating design methods and obtaining synergies, which in turn reduces the enabling effect on the center circle.
F48: How methods were used need to be captured	It was frequently observed that 'how the design method was used' was relevant to capture. Mainly since this has positive second-order effects on the center circle, capturing 'how the design method was used' supported in either highlighting and/or preventing misuse of the design method. In turn, this becomes an enabler towards the center circle.
F49: Who were part of the creating the results need to be captured	It was frequently observed that those who were involved in the design method used are relevant to capture. Capturing who was involved can both ensure that the 'right' people and competence take part in the design method and increase traceability. In turn, this acts as an indirect enabler towards the center circle.
F50: Do not know what or how to capture	It was frequently observed that it is not clear what sustainability-relevant information and data need to be captured. Nor is it clear how that data will be retrieved and stored. In turn, this affects the information used as input to design methods along with its representation and has a direct impact on the center circle.
F51: Design methods transfer understanding of 'how and what' sustainability relevant information and data to capture to practitioners	It was frequently observed that adopting new and improved design methods supported clarifying what information and data need to be captured in <i>sustainable design practices</i> . In turn, this can have several enabling effects on Category H and thus also the adoption of <i>sustainable design practices</i> .
F52: Design methods transfer ' <i>design know-what</i> ' to practitioners	It was constantly observed that adopting new and improved design methods transfers 'design know- what' to practitioners. It more specifically increased their awareness and understanding, i.e., knowledge, of what change of and/or new design practices are needed to transform to sustainable design practices. It highlighted what sustainability-related 'problems' need to be addressed and/or what knowledge is relevant to produce, which in turn can be used to increase the sustainability performance of their solutions. Practitioners had their presumptions about what sustainable design or sustainable design practices imply or mean, and what 'problems', or knowledge is relevant to their company. Adopting the design methods challenged these presumptions and supported them in clarifying what knowledge is relevant, and what sustainability-related 'problems' need to be solved. It was occasionally observed as also transferring the ownership of these 'problems' to the practitioners. Furthermore, the practitioners themselves also frequently claimed to obtain increased awareness and understanding of what change of design practice is needed by adopting the design methods. In turn, this can have several enabling effects on Category D and thus also the adoption of sustainable design practices.
F53: Design methods transfer ' <i>design know-how</i> ' to practitioners	It was constantly observed that adopting new and improved design methods transfers 'design know- how' to practitioners. It more specifically increased their awareness and understanding i.e., knowledge, of how their current design practices can be changed, and ideally should be changed. It highlighted how to solve their sustainability-related 'problems' and/or how to produce relevant knowledge, which in turn can be used to increase the sustainability performance of their solutions. It was constantly observed that this supported practitioners in better understanding of what actions and by whom are required to reach specific outcomes, i.e., how to adopt sustainable design practices. In turn, this can have several enabling effects on Category F and thus also the adoption of <i>sustainable design practices</i> .

3.6. Comparing the proposed descriptive framework with interdisciplinary literature

The proposed framework was lastly compared to interdisciplinary literature to understand how the findings relate to existing theories (Charmaz, 1996), as visualized in Fig. 1. This activity added further nuance and depth to the empirical findings where literature in both the design, and management domain was identified using a snowballing approach (Wohlin, 2014). The focus was to identify literature that can be used to further explain and clarify the empirical findings to bridge these domains and frame the industrial adoption of design methods from all three perspectives (i.e., process and methodology, organization, and human behavior). This, in turn, resulted in a set of what we refer to as systemic barriers and propositions, which aim to make the empirical findings more easily managed and absorbed. The systemic barriers are of a systemic nature that summarizes and highlights key barriers. The propositions are instead our proposed means, or suggested action, for addressing these systemic barriers based on the findings in our study. These aim to guide both research and practice on what actions and pressing issues need to be addressed to enable the adoption of sustainable design practices.

4. Results

The results from the Glaserian grounded theory analysis are divided into two parts. Section 4.1 presents the resulting descriptive framework consisting of 53 interdisciplinary factors that influence the adoption of sustainable design practices using new and improved design methods. Section 4.2 highlights three enabling factors that were captured by the framework, which can expand and nuance the role of new and improved design methods. Furthermore, the empirical findings are further elaborated upon in Section 4.3 where two new concepts are introduced and conceptualized.

4.1. The descriptive framework capturing 53 factors influencing the adoption of sustainable design practices using new and improved design methods

The descriptive framework captures 53 different and interdependent factors that influence the adoption of *sustainable design practices* using design methods. These factors relate to the eight identified core categories and are further described below.

- Category A relates to practitioners' understanding of 'why and how' to appropriately use the design methods proposed by *method experts* and/or *developers*.
- Category B relates to *method developers*' understanding of practitioner needs along with if and how they are translated into the design and development of the new design methods.
- Category C relates to the development of the design methods along with if and how they fit into a company's current design process.
- Category D relates to the nature and characteristics of new *sustainable design practices* along with if and how that, in turn, influences the

practitioners, and the new design methods proposed by *method experts* and/or *developers*.

- Category E relates to *method experts*' understanding of the company case and context along with if and how the new design methods are appropriate and applicable.
- Category F relates to practitioners' engagement and how that influences the use of the new design methods proposed by *method experts* and/or *developers*.
- Category G relates to if and how the new design methods proposed by *method experts* and/or *developers* can achieve synergies and integration with other existing and/or new design methods.
- Category H relates to the information and data in *sustainable design practices* along with how that, in turn, influences the practitioners and the new design methods proposed by *method experts* and/or *developers*.

A summary of all the 53 factors is provided in Table 5, which also indicates each factor's occurrence and degree of influence. The table is best understood together with Fig. 2.

One of the main discoveries among the findings were the identified factors' interdependencies and interdisciplinary nature. The interdependent nature of the framework is further visualized in Fig. 2 where a network diagram is used to highlight the network of factors. The factors are either acting as barriers or enablers, depicted as directed arrows towards the blue center circle, or another core category, depending on the influence (i.e., direct, or indirect). The identified core categories A-H are displayed as yellow boxes. All positive influences, i.e., the enablers, are blue with a plus sign next to the arrow. All negative influences, i.e., the barriers, are red with a minus sign next to the arrow. Several of the factors are similar in nature and phrasing, but each factor results in a unique relation and is therefore also treated as a unique factor. Moreover, the framework also captures and displays factors that indirectly influence the center circle, which occurs when there are factors between categories that, in turn, result in a second and/or third-order effect. For example, F2 ("lack of understanding limits practitioner engagement") is a barrier from Category A to Category F, and it was frequently observed that a lack of understanding of 'why and how' to use methods resulted in decreased engagement by the practitioners. This example highlights the complexity captured by the framework since F2 has negative secondorder effects on F34 ("Practitioners' engagement provides understanding"), an enabler to Category E. In turn, this becomes a barrier to Category G via barrier F32 ("Limited understanding can make integration difficult"). Moreover, Category F is influenced by F20 ("Design methods require practitioners' engagement") which in combination with F2 becomes a barrier to adoption. A full and detailed analysis of all the factors is provided as supplementary information. The most significant parts of the framework will, however, be further highlighted in the coming sub-section. Additionally, Section 5 highlights the interdisciplinary nature of the framework while clarifying the practical and theoretical implications of our findings.

4.2. The enabling effects of new and improved design methods

The analysis identified three main enablers that can clarify and nuance what role design methods can have in supporting practitioners in their organization's *sustainability transformation*. The subsections below complement the descriptions of factors 51, 52, and 53 in Table 5 with examples and a more detailed description.

4.2.1. Factor 51 – design methods transfer understanding of 'how and what' sustainability relevant information and data to capture to practitioners

This factor is a direct enabler of Category H ("Information and data capturing in sustainable product development"). It was frequently observed that adopting new and improved design methods supported clarifying what data needs to be captured in *sustainable design practices*. Design methods have a prescribed input along with how information

and data should be represented, this was observed as supporting practitioners to create a common language and representation of sustainability information and data. One practitioner did, for example, state that the design methods "provide a common language and shared understanding of sustainability" as well as "terminology around sustainability". Furthermore, the design methods also prescribe what sustainability information and data is needed to appropriately use the design methods, i.e., relevant sustainability data. This clarified what information and data is necessary to retrieve and store both from internal functions and external stakeholders, such as suppliers. One practitioner did for example state that "the method resulted in a way to express component requirements to suppliers". Another practitioner stated that "earlier, procurement did not have the right knowledge about sustainability to be able to make decisions (related to suppliers)". In turn, this can have several enabling effects on Category H and thus also the adoption of sustainable design practices.

4.2.2. Factor 52 – design methods transfer 'design know-what' to practitioners

This factor is a direct enabler of Category D ("New sustainable design practices"). It was constantly observed that adopting new and improved design methods transfers 'design know-what' to practitioners. It more specifically increased their awareness and understanding, i.e., knowledge, of what change of and/or new design practices are needed to transform to sustainable design practices. It highlighted what sustainability-related 'problems' need to be addressed and/or what knowledge is relevant to produce that, in turn, can be used to increase the sustainability performance of their solutions. Practitioners had their presumptions about what sustainable design or sustainable design practices imply or mean, and what 'problems', or knowledge is relevant to their company. Adopting the design methods challenged these presumptions and supported them in clarifying what knowledge is relevant, and what sustainability-related 'problems' need to be solved. It was occasionally observed as also transferring the ownership of these 'problems' to the practitioners. One practitioner did for example state that the main purpose of the design methods is to "ask the right questions", i.e., what 'problems' to solve or knowledge to produce. Furthermore, the practitioners themselves also frequently claimed to obtain increased awareness and understanding of what change of design practice is needed by adopting the design methods. One practitioner did, for example, state that "I have learned so much more, I thought I already knew a lot". Another practitioner stated that "it supports creating increased awareness". It was also stated that it supported "shedding light" on sustainability. In turn, this can have several enabling effects on Category D and thus also the adoption of sustainable design practices.

4.2.3. Factor 53 – design methods transfer 'design know-how' to practitioners

This factor is a direct enabler of Category F ("Practitioners' design method engagement"). It was constantly observed that adopting new and improved design methods transfers 'design know-how' to practitioners. It more specifically increased their awareness and understanding i.e., knowledge, of how their current design practices can be changed, and ideally should be changed. It highlighted how to solve their sustainability-related 'problems' and/or how to produce relevant knowledge that, in turn, can be used to increase the sustainability performance of their solutions. The design methods prescribe who needs to be involved and how they should be involved in the design methods. Furthermore, the design methods also prescribed a structured approach for what questions need to be answered as well as what actions and inputs are needed to reach specific outcomes, i.e., how to produce knowledge or how to solve a 'problem'. It was constantly observed that this supported practitioners in better understanding of what actions and by whom are required to reach specific outcomes, i.e., how to adopt sustainable design practices. One practitioner did for example state that the design methods support in a "structured way of how we can achieve



Fig. 2. The final descriptive framework.

the sustainability goals – where to start and then which actions to take". Another practitioner stated that it "provides a structured way of integrating sustainability". In turn, this can have several enabling effects on Category F and thus also the adoption of *sustainable design practices*.

4.3. Conceptualizing the nature of design methods

The literature provided in Section 2.3 was not considered sufficient to explain and clarify several of the identified factors and bridge the three perspectives. Two new concepts referred to as the *situational design problem* and the *dualism of design methods* are therefore proposed and further conceptualized in this section and used in Section 5 to facilitate the discussion.

4.3.1. Introducing the situational design problems of practitioners

From a practitioner's point of view, design methods support solving what was referred to as 'problems' as observed in factors 52 and 53. These 'problems' are situational problems or challenges, and occasionally explicit and unmet needs that arise in the design process when current design practices, or design methods, are considered insufficient. Furthermore, these observed 'problems' are divided into three types or layers: (i) The *Situational design problem*; (ii) *Situational sub-problems*; and (iii) *Contextual problems*. Type one (*situational design problem*) relates to the core of such 'problems' and is typically how problems in the empirical study were and would be, formulated by a practitioner. Examples of such *situational design problems*, or unmet practitioner needs, that arose in the empirical study were:

• How to understand the feasibility and system effects of changing materials?

- · How to assess and realize lifecycle strategies?
- How to capture and define contextually relevant sustainability criteria?
- Need to formulate criteria that can be used in collaboration processes.
- How to assess risks related to long-term sustainability compliance?
- Need a shared understanding of what sustainability means.
- How to identify sustainable materials?
- Need to be able to identify the value of different design strategies.

These situational design problems typically relate to the need, or current challenge, of generating different sets of knowledge considered relevant to the design problem, and this study focused on sets of knowledge required to develop more sustainable solutions. However, to clarify, satisfying such a need or solving a situational design problem by e. g., adopting a new and improved design method does not solve the original design problem as such. Type two (situational sub-problems) were, and would, typically not be stated explicitly by practitioners but are instead sub-problems that have been identified as necessary to solve by e.g., the method developer during the development of the design method. Design methods proposed by e.g., design researchers, are ideally rigorously developed, tested, and sufficiently mature when proposed to practitioners. Situational sub-problems can for example relate to how to structure and represent information in a condensed format such that it can be communicated internally to make the design method user friendly, or how to systematically divide sustainability criteria according to lifecycle phases which might result in a sub-step in the design method. Type three (contextual problems) is important to distinguish from types one and two since these refer to 'problems' that either differ from organization to organization, or from case to case and are thus contextual. Examples of *contextual problems* from the empirical study were requests to adapt the design methods to company language, or what some practitioners referred to as making them "companyfied" and thus differ from organization to organization. Another *contextual problem* instead related to the need to efficiently link the design methods, or rather their accompanied computer tools, to the organization's internal information and data management system. Such systems also differed from organization to organization and became a *contextual problem*. In many instances, *contextual problems* are generally related to explicit requests to simplify, adjust, and/or modify the design methods to better align with their organizational context (e.g., product, internal processes, and resources), or *method ecosystem*.

4.3.2. Introducing the dualism of design methods

The dualism of design methods builds on the finding that factors 52 and 53, where design methods were observed to transfer both 'design know-what' and 'design know-how' to practitioners. The Design know-what encapsulates and transfers knowledge of what situational design problems, situational sub-problems, and occasionally contextual problems, are relevant to practitioners. The Design know-how instead encapsulates and transfers knowledge of how practitioners can solve specific situational design problems, situational sub-problems, and occasionally contextual problems. This also indicates that design methods are constituted by two halves, or a dualism, and is further illustrated in Fig. 3. We use the Double Diamond (Design Council, 2005) for pedagogic reasons to highlight this dual nature and frame design methods as designed artifacts (i.e., designs). The Design know-what is portrayed as and relates to the 'problem space' and the Design know-how to the 'solution space' respectively. As mentioned, the Design Research Methodology (Blessing and Chakrabarti, 2009) is a commonly used research methodology, or 'design process', for developing design methods, and the generic steps of this process are therefore fitted accordingly to clarify this reasoning.

The *dualism of design methods* also supports highlighting that design methods are value-laden artifacts, or designs i.e., design methods are a means to an end, where the end is value-laden. Furthermore, design methods pursue different goals aiming to improve the design process (Blessing and Chakrabarti, 2009) by solving or addressing different *situational design problems*. Furthermore, the *dualism of design methods* highlights and nuances the role of design methods, they can be used to convey and/or transfer:

- (i) How to solve *situational design problems* relevant to the *design problem*.
- (ii) What situational design problems are relevant to the design problem.
- (iii) How to produce knowledge relevant to the design problem.
- (iv) What knowledge about the design problem is relevant to produce.

5. Discussion

The descriptive framework captures and highlights what factors influence the adoption of sustainable design practices using new and improved design methods. However, the mechanisms behind the factors as well as how they should be addressed need to be further discussed and compared to previous research. Furthermore, this enables us to summarize and categorize the 53 influencing factors as nine **systemic barriers** and eight **propositions**, which in turn further highlight and clarify the theoretical and practical implications of our findings.

5.1. Facilitating the adoption of new and improved design methods

Literature in the design domain has mainly treated design methods as prescriptive means for transferring a type of 'know-how' about designing (see e.g., Eder, 1998; Wallace, 2011; Daalhuizen and Cash, 2021; Gericke et al., 2022) and align with what we proposed as the *Design know-how*. Furthermore, we also claim that design methods can, and should, be considered as 'designs' that solve *situational design* problems. Moreover, in our proposed concept we also introduce the *Design know-what* which is less elaborated upon in the design domain. There are similar concepts, such as *method goal* (Daalhuizen and Cash, 2021) and *intended use* (Gericke et al., 2022), but these are not framed nor discussed as potential barriers to adoption. We instead use the *Design know-what* to explicitly highlight design methods as being value-laden artifacts, i.e., not only a means to an end but also an end. Our proposed concept provides a nuanced view of design methods and will be used to describe and explain several of the barriers and enablers captured in the descriptive framework. It is possible to formulate five **systemic barriers** and three **propositions** which are elaborated upon in the coming subsections.

5.1.1. The situational design problem paradox

The nature of a situational design problem shares similar characteristics to design problems i.e., they are also ill-structured, ill-defined, to some extent wicked, and unique (Rittel and Webber, 1973; Archer, 1979; Buchanan, 1992; Simon, 1969; Dorst, 2006; Gericke et al., 2022), see e.g., F10 ("adaptation is required"), and F14 ("the design process differs from company to company"). A paradox, referred to as the situational design problem paradox is therefore introduced here. This paradox claims that design methods, in theory, are limited in their transferability, or applicability, since situational design problems are thus also unique. Such a paradox, in turn, indicates that the use and applicability of proposed design methods are limited since the situational design problem will differ every time. This claim indicates that previously proposed and used design methods are never applicable again. However, such a claim has partly been proven false since many design methods have been proven useful in several different and unique design situations such as Quality Function Deployment (see e.g., Chan and Wu, 2002), or Business Model Generation (Osterwalder and Pigneur, 2010) also referred to as Business Model Canvas in the context of business design. The situational design problem paradox is, however, still a barrier but needs to be better understood. First, the influence of this paradox depends on how relevant and/or 'common' the situational design problem, i. e., its Design know-what, is to practitioners, and if the design methods are externally valid (Säfsten and Gustavsson, 2020). It was observed in the empirical study that this varied among the different design methods proposed, see e.g., F7 ("need to understand how practitioners currently work") or F24 ("difficult to know which design methods are needed"), and influenced adoption since the design methods need to address a situational design problem relevant to the practitioners. This entails Systemic barrier 1a:

The *situational design problem paradox* limits a design method's transferability and applicability to practitioners.

Second, situational design problems are seemingly unique since modifications and adaptations to the design methods were frequently required in the empirical study, see e.g., F11 ("sustainability is considered complex and needs to be simplified") and F10 ("adaptation is required"). Most of the requests were related to what we refer to as contextual problems and these rarely influenced the outcomes. The requests did, however, occasionally require significant modifications to what we refer to as situational sub-problems and such modifications influenced the outcomes to a larger extent. Such modification and adaptation were not straight-forward and required the facilitation of what we refer to as a method expert, mainly due to F3 ("there are difficulties in understanding how sustainability can be simplified") and F22 ("there is limited sustainability relevant expertise"), and F30 ("understanding enables appropriate simplification"). The design researchers, here acting as method experts, had to study the case company and context in detail to make appropriate modifications and contextualization to not influence the design method outcome negatively, see Category E ("Method experts" understanding of company case and context"). This activity was recognized as highly important and appreciated by practitioners, see e.g., F12 ("increases if fitted into the current design process"). Literature has also



Fig. 3. Depicting the dualism of design methods. Each space is further divided into the Situational design problem (L1), Situational sub-problems (L2), and Contextual problems (L3).

highlighted the need for adaptation and contextualization. Eder (1998) did for example state "the method must be adapted to problem and situation, adapted to different kinds of product and peculiarities of the enterprise" while highlighting the importance of individuals in an organization that can carry out such adaptation. Wallace (2011) also highlighted the need for contextual adaptation when stating that a crucial barrier to adoption is when design methods do not fit practitioners' working practices. Gericke et al. (2020) refer to this context as the method ecosystem where design methods need to fit and typically require adaptation. Gericke et al. (2021) also stated that such context is "fuzzy", claiming that contextual adaptation is one of the reasons for low industrial uptake. We argue that adopting a design method is rarely, or ever, a 'plug-and-play' and requires consideration and facilitation when introduced in new contexts. Furthermore, contextual adaptation of design methods also requires an extensive understanding of the company, case, and context, i.e., domain-specific competence, and highlights the need for method experts who can be responsible for such contextual adaptation. This entails Systemic barrier 1b:

The *situational design problem paradox* requires a design method to be modified and contextually adapted to be applicable and accepted by practitioners, which is a time-consuming activity requiring *method experts*.

The above findings provide empirical evidence that situational design problems are similar to design problems since they also appear to be illstructured, ill-defined, to some extent wicked, and unique. It is, however, beneficial to distinguish what type of problem hinders adoption, i. e., if it relates to the situational design problem, situational sub-problems, or contextual problems. Different activities and resources are required, and suitable, depending on which type of problem needs to be addressed. For example, mature and user-friendly computer tools have been highlighted as important for acceptance and adoption in this empirical study and literature (see e.g., Thia et al., 2005; López-Mesa and Bylund, 2011; Gericke et al., 2020). Araujo et al. (1996) even found that practitioners perceived the utility of a design method improved if it was implemented using a computer. This, on the one hand, emphasizes the need for maturing design methods and developing user-friendly tools as it is important for efficient scaling and increasing the potential use inside an organization. However, the empirical study also highlighted that developing such a 'mature enough' and user-friendly computer tool requires contextual problems to be addressed. This is both time-consuming and requires a detailed understanding of the organization, and occasionally requires competencies and resources beyond what a method developer possesses. Wallace (2011) did for example claim that "it is not the main task of academic researchers to write software code, even if they are competent at it ... the resulting software is not likely to be sophisticated, robust or user-friendly by modern standards". We, therefore, argue that some problems hindering adoption are better addressed by e.g., method experts, and occasionally require support from the organizations seeking to adopt the design method. Furthermore,

design methods are commonly requested to be modified and adapted, or 'improved', to fit an organization's current design practices and the needs of practitioners. We do, however, argue that it is not always and necessarily the design method, i.e., the *Design know-how*, that is the reason for limited adoption. It is important to understand what *situational design problem* the design method intends to solve i.e., the *Design know-what*, since it can be the main reason for low applicability. This entails **Proposition 1a:**

Method experts and/or developers and practitioners need to understand the barriers towards adoption in regards to the *situational design problem, situational sub-problems,* and *contextual problems,* such that resources are allocated effectively.

5.1.2. The dualism of design methods paradox

It became evident in the descriptive framework that there is an initial hesitation or skepticism towards adopting new and improved design methods, and there is a clear preference to keep practices to what we refer to AS IS, see e.g., F37 ("practitioner engagement limited to AS IS"). Practitioners also frequently stressed the need to understand why new and improved design methods are needed, see factors related to Category A ("Practitioners' understanding of 'why and how' to appropriately use design methods"). Several practitioners were hesitant to adopt new design methods before their first use but did however perceive their relevance after the design methods had been tested and used, which becomes a 'chicken or the egg' dilemma. The design methods did address relevant situational design problems, but the practitioners did not consider the situational design problems relevant before using or testing the design methods. This introduces the dualism of design methods paradox which claims that the Design know-what must be intrinsically adopted before the Design know-how can be adopted, but the Design know-what can only be intrinsically adopted by using the design method, i.e., adopting the Design know-how. This is also why we argue that it is important to consider and understand design methods as value-laden artifacts, i.e., not only a means to an end but also an end, which practitioners must intrinsically possess. To clarify, when we propose new and improved design methods to practitioners, we assume either and/or that the practitioner wants: (a) the situational design problem to be solved; and/or (b) to produce that knowledge about the design problem. And they usually do want (a) and/or (b) but the challenge is that they do not know it, 'yet'. This entails Systemic barrier 1c:

The *Design know-what* must be transferred before the need for the *Design know-how* is perceived, but the *Design know-what* is seldom adopted before adopting the *Design know-how* which introduces the *dualism of design methods paradox*.

A similar issue was also reported by Araujo et al. (1996) who conducted a large quantitative study with companies in the United Kingdom and concluded that "many companies are unaware of the potential quality benefits of available methods". This issue and paradox pose a real challenge to adoption, but our empirical study did, however, indicate via factors 52 and 53 that the *situational design problem*, or 'value', of the design method can be transferred when design methods are adopted (i. e., used). Furthermore, López-Mesa and Bylund (2011) also claimed that "the value is only understood when you have some practical experience with it" which supports our paradox and our claim that design methods can transfer the *situational design problem* they intend to solve. This entails **Proposition 1b**:

The *Design know-what* needs to be fully understood or tested before deeming the design method to be of no relevance to the practitioner and their *design problem*. Adopting new and improved design methods can transfer what *situational design problems* and what set of knowledge are relevant to their *design problem*.

A. Mallalieu et al.

"We all love to instruct, though we can teach only what is not worth knowing" – Jane Austen in *Pride and Prejudice* (1813, chapter 4)

5.1.3. The prescriptive nature of design methods

'Change' is in general acknowledged as a challenging social phenomenon and it is clear how it influences the adoption of new and improved design methods since it will naturally require process and methodological change. López-Mesa and Bylund (2011) are one of few in the design domain who acknowledge that "people do not want to change their way of working" while briefly referring to Kotter (1995). Similar was highlighted by Jagtap et al. (2014) who also referred to Kotter (1995), but this did not gain any significant traction in the design domain. It is important to acknowledge change which results in **Systemic barrier 1d:**

Adopting new and improved design methods induces process and methodological change which is a well-studied social phenomenon and acts as a natural barrier to adoption.

Design methods are in their nature a prescription of how a *situational design problem* should be solved and, in turn, affect the autonomy of practitioners since it limits their ability to influence how they design. *Autonomy* is a critical element in achieving social acceptance and motivation (Hackman and Oldham, 1976; Deci et al., 2017) and thus requires attention. Furthermore, López-Mesa and Bylund (2011) also found that practitioners believe that "methods reduce freedom to think and are boring" but did not specifically refer to autonomy. The role of autonomy is important to highlight and acknowledge since it can provide further clarification of why there is an instant hesitation to adoption and why practitioners frequently want to be involved in modifying and adapting the design methods, see Section 2.4 or Hackman and Oldham (1976). This entails **Systemic barrier 1e:**

The nature of adopting prescriptive design methods points to reduced autonomy which is a key element in work design and more specifically the *Self-Determination Theory*.

Factor 52, or the Design know-what, was observed to convey what change is required and we propose it be utilized to facilitate change and social acceptance. The Design know-what is a potential means for transferring increased awareness and understanding and aligns with the first step provided in Kotter's change model, i.e., "establishing a sense of urgency" (Kotter, 1995). Furthermore, design methods, or the Design know-how provide a prescription for how to realize this change. However, as evident in sub-sections 5.1.1 and 5.1.2, adoption is challenging and seldom a 'plug-and-play'. It is therefore worth considering whether we want the practitioner to either: (a) solve the situational design problem using the design method as it is exactly prescribed; or (b) solve the situational design problem that the design method intends to solve. Mainly since (b) is a probable superset to (a) but not vice versa. We argue that (b) is more important than (a) since the design method is one of many ways to solve the situational design problem (Rittel and Webber, 1973; Buchanan, 1992). Furthermore, we also argue, due to the contextual nature of designing, that a practitioner is best suited to deem how to modify and use a design method, assuming the practitioner is fully knowledgeable of the Design know-what. This entails Proposition 1c:

Design method adoption should be carried out in a cyclic or iterative nature. First-cycle adoption is an initial trial of the *Design know-how* increasing awareness and understanding via the *Design know-what* i. e., what change is needed. Second-cycle adoption focuses on fully adopting the *Design know-what*, either via the *Design know-how* or by modifying and adapting it to better suit the *situational design problem* and its context.

5.2. The adoption of new and improved design methods in the context of a sustainability transformation

Section 5.1 focused on providing a better understanding of why it is natural to encounter challenges when proposing new and improved design methods to practitioners, and how adoption can be facilitated. Section 5.2 instead shifts focus and aims to highlight why product development and manufacturing organizations must face, and can face, these challenges to achieve a *sustainability transformation*. It is possible to formulate four **systemic barriers** and five **propositions** which are elaborated upon in the coming subsections.

5.2.1. Organizational change in the paradigm of sustainable design

The descriptive framework includes several barriers that relate to how design or rather what we introduce here, a paradigm of product design, is practiced in industry from a process and methodological perspective. This paradigm of product design represents a persisting way of practicing design in product development and manufacturing organizations, which in turn pose significant challenges to practitioners and their organization's sustainability transformation, see e.g., F16 ("radical changes limit the ability to fit the current design process"), F27 ("requires collaboration with new internal and external actors"), and factors related to Category D ("New sustainable design practices"). The paradigm of product design mainly perceives and treats product design as functionality and performance in use (e.g., weight, power), geometry, component, and part integration, along with material selection. Such a paradigm of product design can also be found in literature. For example, Cross (2023) also refers to the traditional design as more focused on physical objects, and Ulrich and Eppinger (2016, p3) provide the following description:

"The design function plays the lead role in defining the physical form of the product to best meet customer needs. In this context, the design function includes engineering design (mechanical, electrical, software, etc.) and industrial design (aesthetics, ergonomics, user interfaces)."

Lee (2021) criticizes current design practices, while referring to the paradigm of design thinking, arguing that it still withholds a focus on making physical artifacts despite other claims. We argue that such a narrow view of design results in a limited ability to implement sustainable design practices and propose sustainable solutions i.e., incorporating the eight socio-ecological principles throughout the product's full lifecycle. The paradigm of product design results in a systemic process and methodological incompatibility since the nature of the new and improved design methods are incompatible with how design, or rather product design, is currently practiced in these organizations. Furthermore, this influenced both the ability and attitude towards adopting the proposed design methods and did in several instances require significant modification, which in turn influenced the method outcomes negatively. Examples relate to the inclusion of social aspects in the full value chain and can pose a risk of preserving social issues such as inequality and health risks during material extraction. There was in general a limited ability to consider the full lifecycle of products, see e.g., F4 ("the concept of lifecycles is difficult to understand") and F19 ("sustainability principles are not well understood"). This means that not only social but also ecological issues can be preserved in downstream and/or upstream activities. Furthermore, Mallalieu et al. (2023) also highlight this issue and provide concrete examples of how sustainability criteria can be incorporated into design space explorations while highlighting the difficulties in integrating less conventional sustainability criteria. Physical aspects are compared to non-physical aspects more easily treated in current design practices. This entails Systemic barrier 2a:

Organizations' current design practices align with the *paradigm of product design* and result in a systemic process and methodological incompatibility with *sustainable design practices*.

Our claim is supported by and elaborated on by other scholars, where Eppinger (2011) for example criticized current "product design practices" (i.e., the paradigm of product design) stating that in terms of incorporating sustainability it "remains largely in the dark ages". Pigosso et al. (2013) did not explicitly state the need for systemic changes to current design practices but did for example state that companies must "implement life cycle thinking" to enable the development of more sustainable solutions. Bocken et al. (2014) stated that "these types of changes require a fundamental shift in the purpose of business and almost every aspect of how it is conducted" when discussing strategies to propose more sustainable solutions. Ceschin and Gaziulusoy (2016) also emphasize the need to go from product design to "systemic design". Sumter et al. (2018) also argue that the "role of product design will, and must, expand". Julier and Kimbell (2019) focus on social design, i.e., design aimed at addressing social issues, and they question if current design practices are sufficient to address such complex sustainability-related design problems. Hallstedt et al. (2020) elaborate on the future of design and designers accordingly:

"These trends clearly impact the role of the designers that are expected to design entire solutions as opposed to merely artefacts in the future. This implies that designers need to consider not only the product performance and cost, but also the behavior of products and solutions and their impact over complete life-cycles, developed and organized by business networks together with several suppliers and other partners with different capabilities."

The *paradigm of product design* thus provides a limited scope to design which is not sufficient to produce relevant and required knowledge about the *design problem* to develop and propose sustainable solutions. We, therefore, propose that the design practices in organizations must adapt and change to instead enter a *paradigm of sustainable design*. This entails **Proposition 2a:**

There is a need for a systemic shift away from the *paradigm of product design* towards the *paradigm of sustainable design* to enable the adoption of *sustainable design practices*.

Several of the identified barriers relate to social acceptance and/or an attitude towards conflicts with current design practices, i.e., the need for change. This explains the explicit requests to modify and adapt proposed design methods to better fit their current design practices, see e.g., F6 ("need to understand how design methods can be fitted into the current design process"), F10 ("adaptation is required"), and F12 ("increases if fitted into the current design process"). Literature in the design domain (see e.g., Eder, 1998; Wallace, 2011; Booker, 2012) also emphasizes the need to modify and adapt design methods to fit into current design practices without referring to managerial literature. Furthermore, Faludi et al. (2020) in the context of sustainable design practices also stated that "tools should be easier to apply and compatible with existing business and design methods and processes". Quella and Schmidt (2003) also highlighted the importance of integrating environmental strategies with the current design process along with ensuring "social acceptance". This is according to us a pragmatic and natural mindset or approach which method experts and/or developers tend to utilize to increase the chances of adoption, i.e., they reduce the need for change since change typically is challenging to achieve (Samuelson and Zeckhauser, 1988; Kotter, 1995). A similar approach is utilized and proposed in several other studies (see e.g., Lindahl, 2006; O'Hare et al., 2010; Faludi et al., 2020; Parolin et al., 2024). However, such a mindset and 'pragmatic approach' can result in stagnation and limited ability to adopt sustainable design practices and ability to develop and propose more sustainable solutions. This entails Systemic barrier 2b:

There is a pragmatic mindset in the *paradigm of product design* where the attitude and social acceptance towards process and methodological conflicts, i.e., the need for change, limits the ability to appropriately adopt *sustainable design practices*.

Entering the paradigm of sustainable design will instead require a shift of mindset towards the adoption of sustainable design practices using new and improved design methods. Transitioning to such a paradigm will result in process and methodological conflicts, i.e., there will be a need for change. We therefore argue that both practitioners, method experts, and/or developers need to consider the human-behavioral aspects proactively and intentionally to achieve such change. Furthermore, Hackman and Oldham (1976) highlight autonomy and the importance of practitioner involvement when designing work tasks to achieve motivation and social acceptance. Kotter (1995) also urges to involve people in the process of organizational change to gain traction. A similar proposal was made by Faludi et al. (2020) but without referring to managerial literature that explicitly highlights co-creation. Co-creating, or co-evolving current design practices requires a change of mindset by both method experts and/or developers, and practitioners, and do for example require two-way communication. Potential conflicts and incompatibility can for example serve to highlight where current design practices need to evolve to better align with sustainable design practices. Adaptation, modification, or pragmatic requests, by practitioners need to be considered. However, practitioners must at the same time understand that a systemic change is needed to enable a sustainability transformation. For example, F44 ("limited when there is lack appropriate data") both poses a barrier but also indicates that there is a need for making such data accessible. Furthermore, testing and using new and improved design methods was also observed to improve current design methods and practices and incorporate sustainability, see e.g., F40 ("can integrate sustainability into other design methods"). This entails Proposition 2b:

The mindset in the *paradigm of sustainable design* requires a change of attitude and increased social acceptance towards process and methodological conflicts, i.e., the need for change, and is seen as an opportunity to improve and co-evolve towards *sustainable design practices*.

5.2.2. Pseudo-sustainability and the need for cognitive bias reduction

Several of the barriers identified in the descriptive framework can be related to insights provided by literature treating agency theory (Mitnick, 1992; Eisenhardt, 1989), institutional theory (Meyer and Rowan, 1977; DiMaggio and Powell, 1983), and cognitive barriers or biases (Simon, 1969; Mitnick, 1992). We argue that there is a presence of several different cognitive biases that currently influence adoption negatively, see e.g., F3 ("there are difficulties in understanding how sustainability can be simplified"), F4 ("the concept of lifecycles is difficult to understand"), F19 ("sustainability principles are not understood"), F25 ("there is a need to train and educate designers within sustainable design"), and F37 ("practitioner engagement limited to AS IS"). Furthermore, Klotz et al. (2018) studied how different cognitive biases can lead to 'sub-optimal sustainability performance' and broadly define cognitive biases as "systematic patterns of deviations from classical notions of rationality are called 'cognitive biases', and influence what we view as desirable and possible". Weber (2017) instead discusses cognitive barriers in decision-making and how they result in a "status quo" that limits sustainable development and long-term benefits. The cognitive biases in our study mainly arise due to competence and presumptions of sustainable design practices, different self-interests, personal preferences, and/or social norms. The need to use *heuristics* also appears in the descriptive framework, see e.g., F11 ("sustainable design is considered complex and needs to be simplified"). The presence of these cognitive biases provides a further explanation for the frequent requests by practitioners to modify, adapt, and occasionally simplify, the proposed design methods to better fit their current design practices, which in turn results in the 'pragmatic approach' utilized in the design domain discussed in Section 5.2.1. Design methods are, as previously highlighted, in many cases treated as insufficient by practitioners and in need of 'improvement' to better fit AS IS, or status quo due to cognitive biases.

A. Mallalieu et al.

However, such a 'pragmatic approach' to account for *cognitive biases* can, in turn, lead to a state of *pseudo-sustainability* i.e., the adoption of instances, or 'cherry-picked' parts, of *sustainable design practices. Pseudo-sustainability* is the result of either the intentional, or unintentional, exclusion of any of the eight socio-ecological principles in either of a product's lifecycle phases, which results in *pseudo-sustainable designs*. This entails **Systemic barrier 2c**:

The presence of *cognitive biases* influences the ability to fully embrace and adopt *sustainable design practices* using new and improved design methods, which in turn leads to a state of *pseudo-sustainability*.

The dualism of design methods paradox introduced in Section 5.1.2 is also relevant to discuss here. We there claimed that design methods are value-laden and that a method expert and/or developer thus make assumptions when proposing a new and improved design method to a practitioner, i.e., that the practitioner wants: (a) the situational design problem to be solved; and/or (b) to produce that knowledge about the design problem. And that they usually do want (a) and/or (b) but the challenge is that they do not know it, 'yet'. The cognitive biases of practitioners i.e., their preferences, self-interests, current competence, and presumptions towards sustainable design practices, limit them from currently seeing the need of (a) and/or (b), i.e., practitioners do not intrinsically possess the end of the proposed design methods. This barrier mainly relates to the Design know-what and not the Design know-how of the proposed design methods. Cognitive biases must therefore be systematically challenged in product development and manufacturing organizations since they will continue to influence adoption negatively irrespective of how 'good' or 'mature' proposed new and improved design methods are.

5.2.3. Introducing sustainable design thinking

There has been extensive research about *sustainable design practices* and what remains is thus the actual implementation. Blizzard and Klotz (2012) and Bocken et al. (2014) for example propose generic strategies for how product development and manufacturing organizations can improve. There are also more extensive frameworks focusing on how organizations can mature towards more *sustainable design practices*, see e. g., Ceschin and Gaziulusoy (2019) and Pigosso et al. (2013). There is also a plethora of underutilized design methods (Faludi et al., 2020). Baldassarre et al. (2020) also claimed:

"However, these theoretical speculations on sustainable design will not go to great lengths, unless they are tied to solid business considerations. Indeed, evidence shows that sustainable design ideas can be implemented successfully only when they are grounded into the objectives and operations of organizations."

For example, practitioners on lower levels have contextual awareness and make operational decisions with tangible impact on the design, or product. There is, however, also a need to understand and practice sustainable design top-down to ensure appropriate managerial support (see e.g., Bey et al., 2013), by e.g., supporting change, defining aligned targets and objectives, as well as allocating resources to enable such practices. This study also indicates that a *sustainability transformation* will be challenging unless we systematically challenge the *cognitive biases* of practitioners at every level of an organization (and potentially society). *Sustainable design* must be understood and practiced on all levels of an organization, or what Eisenhardt (1999) refers to as a *collective intuition* i.e., a common understanding or collective competence. The empirical findings highlight the specific potential of design methods (see factor 52 "design methods transfer '*design know-what*' to practitioners"), to challenge practitioners' *cognitive biases*, e.g., values and norms, which currently hinder the adoption of *sustainable design practices*. Similar potential was also reported by Liedtka (2015) where *Design thinking*, or *designerly ways of knowing, thinking, and acting* (Cross, 2023), was shown to address different kinds of *cognitive biases*. There is thus a need for a supposed *Sustainable design thinking*² that every practitioner in a product development and manufacturing organization can practice. This entails **Proposition 2c:**

The *cognitive biases* of practitioners can be challenged by practicing *Sustainable design thinking* on all levels of an organization.

The proposed concept of Sustainable design thinking requires further investigation and development, and the paradigm of design thinking (Brown, 2008; Brown and Katz, 2011; Verganti et al., 2021) can serve as a role model in how it has effectively gained traction in different problem-solving practices by transferring the core principles of design (Dorst, 2011; Cross, 2023). Furthermore, two key competencies need to be transferred and thus incorporated into Sustainable design thinking: (1) Design competencies, which refer to the principles of designerly ways of knowing, thinking, and acting which for example focus on the co-evolution of problem-solution and a better understanding of the design problem (see e.g., Cross, 1982; Dorst and Cross, 2001; Dorst, 2011; Cross, 2023). (2) Sustainable design competencies, which refer to the principles presented in the Framework for Strategic Sustainable Development (Broman and Robert, 2017). Moreover, Sustainable design thinking is not striving to disregard or substitute the role of e.g., sustainability experts, or designers with specialized knowledge in sustainable design in organizations. It is neither striving to substitute proven frameworks and design methods for sustainable design, such as the design methods used in the empirical study, or previously proposed design methods within eco-design, design for sustainability, and/or quantitative design methods (e.g., LCA). Sustainable design thinking mainly aims to develop base competence that can and needs to, be practiced by any designer,³ or sustainable designer,⁴ in organizations to systematically challenge the cognitive biases present in organizations. The paradigm of sustainable design can only be entered when Sustainable design thinking is collectively understood and accepted in an organization.

"The transition between competing paradigms cannot be made a step at a time, forced by logic and neutral experience. Like the gestalt switch, it must occur all at once (though not necessarily in an instant) or not at all." –

Thomas Kuhn (1970)

5.2.4. Information and data management in design methods

There are several upcoming legislations such as the Digital Product Passport, Corporate Sustainability Reporting Directive, and EU taxonomy. This entails the need for sufficient and mature information and data within product development management capabilities and manufacturing organizations, such as accessibility to data to ensure traceability throughout the product's lifecycle. Furthermore, Rashid et al. (2013), also highlighted that resource-efficient, or circular, designs require extensive information and data exchange between organizational functions, and across the value chain. The descriptive framework, however, frames several barriers related to organizations' current information and data management capabilities, see core Category H ("Information and data capturing in sustainable design practices") and its related factors. The findings indicate that the current information and data management capabilities are insufficient to meet the increased demands that entail upcoming legislation and sustainable design practices.

² Shapira et al. (2017) tried to define such a *Sustainable design thinking*, but this proposal has not progressed since and it was neither focused on cognitive biases, nor proposed to be used on every level of an organization.

³ "Everyone designs who devises courses of action aimed at changing existing situations into preferred ones" Simon (1969).

⁴ "If you try to ensure long-term human well-being within the limits of the natural world, then you design for sustainability" Klotz et al. (2018).

This issue has been raised in previous literature (see e.g., Gmelin and Seuring, 2014; Chebaeva et al., 2021; Diaz et al., 2021). Parolin et al. (2024) also highlight the issue of insufficient real data for early phase assessments, but propose to "use real data if possible". Booker (2012) argued that a performance metric for design methods should be to what extent it utilizes available data, implying there is typically a lack of data. López-Mesa and Bylund (2011) also claim that new and improved design methods "require information that is not available in practice" along with a "lack of agreed vocabulary and taxonomy". Borsato (2014) refers to this barrier and introduces "ontology building", i.e., data and information that are represented differently need to be bridged to enable desired analyses. Mallalieu et al. (2022) also found that data and information incompatibility among different design methods can limit appropriate adoption. The above examples align with F44 ("Limited when there is lack of appropriate data") and F45 ("The information captured needs to be understood"). Furthermore, the empirical findings also indicated a misalignment between what practitioners perceived as possible, and what is practically feasible with the organization's current information and data management capabilities, see F50 ("Do not know what or how to capture"). Trevisan et al. (2023) also found a misalignment in their study, stating that IT departments had an insufficient understanding of sustainability, and the sustainability departments had insufficient competence related to IT. Schöggl et al. (2023), similarly, found discrepancies in how organizations align their digitalization and sustainability efforts among departments. This poses a risk where 'theoretically sustainable' designs are proposed, but infeasible in practice, resulting in pseudo-sustainable designs. This entails Systemic barrier 2d:

Product development and manufacturing organizations' current information and data management capabilities are insufficient to adopt *sustainable design practices*. There is currently a misalignment between departments of what is practically feasible from an information and data management perspective, which risks leading to the proposal of *pseudo-sustainable designs*.

New and improved design methods can support improving the information and data management capabilities, see F51 ("design methods transfer understanding on 'how and what' sustainability relevant information and data to capture"). Design methods can, for example, be used to streamline terminology and how to structure information and data. Design methods can also clarify what information and data are required for different designs, supporting cross-functional collaboration by e.g., aligning expectations and clarifying practical feasibility between design, IT, and sustainability departments. Schöggl et al. (2023) also call for increased collaboration between IT and sustainability departments, and this study has shown that design methods can serve as effective means to facilitate such cross-functional collaboration. Furthermore, design methods have also been claimed as effective means to improve accountability, record keeping, and traceability (Araujo et al., 1996; Eder, 1998; Eder, 2009), which aligns with our study (see e.g., F48 "How methods were used need to be captured", and F49 "Who were part of the creating the results need to be captured"). This can, in turn, support organizations meeting upcoming legislation. Altogether, this entails **Proposition 2d:**

Adopting new and improved design methods can support product development and manufacturing organizations to improve their information and data management capabilities, enabling more sustainable designs.

There is also a growing body of literature that focuses on emerging digital technologies that can support addressing some of the identified barriers related to information and data management capabilities (see e. g., Rusch et al., 2023; Han et al., 2023; Neri et al., 2023). Examples of emerging technologies include Internet of Things, blockchain technology, and big data analytics. Apart from emerging technologies, Rusch et al. (2023) and Schöggl et al. (2024b), promote inter-organizational

collaboration combined with digital technologies to improve transparency and traceability (among other things) in the value chain. Trevisan et al. (2023), similarly, propose the concept of a circular ecosystem, which (among other things) aims to increase transparency and data sharing among collaborating actors in the value chain. Increased interaction and collaboration between actors in the value chain can be expected to enable the realization of more circular, and potentially sustainable designs (Schöggl et al., 2023; de Vasconcelos Gomes et al., 2024), but such collaboration has also been highlighted as challenging (see e.g., Schöggl et al., 2024a). Moreover, design methods have the potential to support such collaborative efforts, similar to how they can support internal cross-functional collaboration from an information and data management perspective. This entails **Proposition 2e**:

Adopting new and improved design methods can facilitate collaboration between actors in the value chain from an information and data management perspective, enabling more sustainable designs.

6. Conclusion

This paper reports on an empirical study focusing on the adoption of sustainable design practices using design methods. Three product development and manufacturing organizations have been involved in multiple case studies, where qualitative data was collected using participant observation. Previous studies have provided several insights related to this topic, but the literature is scattered and divided across what we refer to as the design and management domains. This poses a research gap. This study therefore utilized an explorative and interdisciplinary approach using Glaserian grounded theory analysis and addressed the following research question: "What influences the adoption of sustainable design practices using new and improved design methods?". The research question is answered by identifying 53 interdisciplinary factors captured in a descriptive framework, which includes both process and methodological, organizational, and human-behavioral aspects. This study thus addresses the identified research gap. The empirical findings are also discussed and compared to interdisciplinary literature to further clarify and explain them. This results in nine systemic barriers and eight propositions that contribute to both theory and practice, which serve as our recommended actions and key issues to address by both research and practice. The key points are summarized below:

- Design methods are elaborated upon, where two new concepts referred to as the *dualism of design methods* and the *situational design problem* are introduced to better explain the identified barriers and enablers.
- Design methods address, or solve, *situational design problems* that share characteristics with *design problems*. This entails the *situational design problem paradox*, which limits a design method's applicability, highlighting that design methods' relevance varies, and their adoption always requires modification and adaptation.
- Practitioners are unable to see the value, or purpose, of new and improved design methods before they are adopted, which entails the *dualism of design methods paradox*.
- Design methods are prescriptive in their nature, which in turn results in natural human-behavioral barriers to their adoption.
- The process and methodological nature of product development and manufacturing organizations' current design practices are framed as a *paradigm of product design*. This entails the need for systemic organizational changes in such organizations to better align with *sustainable design practices*.
- Practitioners are urged to enter the *paradigm of sustainable design*, which requires a change of mindset, or attitude, with an increased social acceptance towards change.
- The presence of cognitive biases is highlighted and how it risks leading to an undesired state of *pseudo-sustainability*.

- The cognitive biases inside organizations must be challenged on all levels to enable the adoption of *sustainable design practices*.
- *Sustainable design thinking* is proposed and outlined as a potential enabler that aims to develop a base competence in sustainable design to systematically challenge the cognitive biases present in organizations.
- The information and data management capabilities of product development and manufacturing organizations limit the ability to adopt *sustainable design practices*.
- Design methods can facilitate collaboration and improve the information and data management capabilities of a value chain.

This research thus paves the way for future research with an interdisciplinary focus to further bridge the gap between the process and methodological, organizational, and human-behavioral perspectives. Future research should focus on assessing the validity of the propositions we provide in Sections 5.1 and 5.2. Five main topics require further attention specifically by future research: First, the role of cognitive biases in industry needs to be further studied as it risks leading to a state of pseudo-sustainability. Second, Sustainable design thinking requires further development and refinement such that it can be tested, evaluated, and used by practitioners. Third, information and data management in design methods is also a research topic that requires further attention. Fourth, design researchers and practitioners need prescriptive support in how to bridge the contextual gap, i.e., what we refer to as *contextual problems*, such that proposed design methods can be appropriately adapted to meet the practitioners' needs. Finally, the paradigm of product design would benefit from further verification by future empirical studies.

CRediT authorship contribution statement

Adam Mallalieu: Writing – review & editing, Writing – original draft, Visualization, Validation, Project administration, Methodology, Investigation, Formal analysis, Data curation, Conceptualization. Sophie Isaksson Hallstedt: Writing – review & editing, Writing – original draft, Validation, Supervision, Resources, Funding acquisition, Conceptualization. Ola Isaksson: Writing – review & editing, Writing – original draft, Supervision, Resources, Funding acquisition, Conceptualization. Matilda Watz: Writing – review & editing, Writing – original draft, Validation, Supervision, Resources, Investigation, Data curation, Conceptualization. Lars Almefelt: Writing – original draft, Visualization, Validation, Supervision, Conceptualization.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

This work has been financially supported by the Sweden's Innovation Agency (Vinnova). Authors also wish to acknowledge all the anonymous participants in the study.

Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.spc.2024.08.023.

References

- Åge, L.J., 2011. Grounded theory methodology: positivism, hermeneutics, and pragmatism. Qual. Rep. 16 (6), 1599–1615.
- Appelbaum, S.H., Habashy, S., Malo, J.L., Shafiq, H., 2012. Back to the future: revisiting Kotter's 1996 change model. J. Manag. Dev. 31 (8), 764–782.

- Araujo, C.S., 2001. Acquisition of Product Development Tools in Industry: A Theoretical Contribution. Technical University of Denmark, Lyngby, Denmark.
- Araujo, C.S., Benedetto-Neto, H., Campello, A.C., Segre, F.M., Wright, I.C., 1996. The utilization of product development methods: a survey of UK industry. J. Eng. Des. 7 (3), 265–277.
- Archer, B., 1979. Design as a discipline. Des. Stud. 1 (1), 17-20.
- Asimow, M., 1962. Introduction to Design. Prentice Hall.
- Auernhammer, J., Roth, B., 2021. The origin and evolution of Stanford University's design thinking: from product design to design thinking in innovation management. J. Prod. Innov. Manag. 38 (6), 623–644.
- Baldassarre, B., Keskin, D., Diehl, J.C., Bocken, N., Calabretta, G., 2020. Implementing sustainable design theory in business practice: a call to action. J. Clean. Prod. 273, 123113.
- Bengtsson, M., Alfredsson, E., Cohen, M., Lorek, S., Schroeder, P., 2018. Transforming systems of consumption and production for achieving the sustainable development goals: moving beyond efficiency. Sustain. Sci. 13, 1533–1547.
- Bey, N., Hauschild, M.Z., McAloone, T.C., 2013. Drivers and barriers for implementation of environmental strategies in manufacturing companies. CIRP Ann. 62 (1), 43–46. Bhander, G.S., Hauschild, M., McAloone, T., 2003. Implementing life cycle assessment in
- product development. Environ. Prog. 22 (4), 255–267. Blessing, L.T.M., Chakrabarti, A., 2009. DRM, a Design Research Methodology. Springer, London.
- Blizzard, J.L., Klotz, L.E., 2012. A framework for sustainable whole systems design. Des. Stud. 33 (5), 456–479.
- Bocken, N.M., Short, S.W., Rana, P., Evans, S., 2014. A literature and practice review to develop sustainable business model archetypes. J. Clean. Prod. 65, 42–56.
- Booker, J., 2012. A survey-based methodology for prioritising the industrial implementation qualities of design tools. J. Eng. Des. 23 (7), 507–525.
- Borgue, O., Paissoni, C., Panarotto, M., Isaksson, O., Andreussi, T., Viola, N., 2021. Design for test and qualification through activity-based modelling in product architecture design. J. Eng. Des. 32 (11), 646–670.
- Borsato, M., 2014. Bridging the gap between product lifecycle management and sustainability in manufacturing through ontology building. Comput. Ind. 65 (2), 258–269.
- Broman, G.I., Robèrt, K.H., 2017. A framework for strategic sustainable development. J. Clean. Prod. 140, 17–31.
- Brown, T., 2008. Design thinking. Harv. Bus. Rev. 86 (6), 84.
- Brown, T., Katz, B., 2011. Change by design. J. Prod. Innov. Manag. 28 (3), 381-383.

Buchanan, R., 1992. Wicked problems in design thinking. Des. Issues 8 (2), 5–21. Bunge, M., 1966. Technology as applied science. In: Contributions to a Philosophy of

- Technology: Studies in the Structure of Thinking in the Technological Sciences. Springer Netherlands, Dordrecht, pp. 19–39.
- Byggeth, S., Hochschorner, E., 2006. Handling trade-offs in ecodesign tools for sustainable product development and procurement. J. Clean. Prod. 14 (15–16), 1420–1430.
- Cambridge, 2024. Adoption (starting to use) English meaning. Retrieved Aug 23, 2024, from https://dictionary.cambridge.org/dictionary/english/adoption.
- Ceschin, F., Gaziulusoy, I., 2016. Evolution of design for sustainability: from product design to design for system innovations and transitions. Des. Stud. 47, 118–163.
- Ceschin, F., Gaziulusoy, İ., 2019. Design for Sustainability: A Multi-level Framework from Products to Socio-technical Systems. Routledge
- Chan, L.K., Wu, M.L., 2002. Quality function deployment: a literature review. Eur. J. Oper. Res. 143 (3), 463–497.
- Charmaz, K., 1996. The search for meanings grounded theory. In: Smit, J.A., Harré, R., Van Langenhove, L. (Eds.), Rethinking Methods in Psychology. Sage Publications, London, pp. 27–49.
- Chebaeva, N., Lettner, M., Wenger, J., Schöggl, J.P., Hesser, F., Holzer, D., Stern, T., 2021. Dealing with the eco-design paradox in research and development projects: the concept of sustainability assessment levels. J. Clean. Prod. 281, 125232.
- Creswell, J.W., 2014. Research Design: Qualitative, Quantitative, and Mixed Methods Approaches. Sage publications.
 - Cross, N., 1982. Designerly ways of knowing. Des. Stud. 3 (4), 221-227.
 - Cross, N., 1992. Research in Design Thinking. Delft University Press, pp. 3-10.
 - Cross, N., 2000. Engineering Design Methods: Strategies for Product Design. John Wiley & Sons.
 - Cross, N., 2023. Design thinking: what just happened? Des. Stud. 86.
 - Cutcliffe, J.R., 2000. Methodological issues in grounded theory. J. Adv. Nurs. 31 (6), 1476–1484.
 - Daalhuizen, J., Cash, P., 2021. Method content theory: towards a new understanding of methods in design. Des. Stud. 75, 101018.
 - de Vasconcelos Gomes, L.A., Homrich, A.S., Facin, A.L.F., Silva, L.E.N., Castillo-Ospina, D.A., Trevisan, A.H., de Carvalho, M.M., 2024. Enablers for circular ecosystem transformation: a multi-case study of Brazilian circular ecosystems. Sustain. Prod. Consump. 49, 249–262.
 - Deci, E.L., Olafsen, A.H., Ryan, R.M., 2017. Self-determination theory in work organizations: the state of a science. Annu. Rev. Organ. Psych. Organ. Behav. 4, 19–43.
 - Design Council, 2005. The double diamond [Online]. Available at https://www.designco uncil.org.uk/our-resources/the-double-diamond/ (2023-11-17).
 - Diaz, A., Schöggl, J.P., Reyes, T., Baumgartner, R.J., 2021. Sustainable product development in a circular economy: implications for products, actors, decisionmaking support and lifecycle information management. Sustain. Prod. Consump. 26, 1031–1045.
 - DiMaggio, P.J., Powell, W.W., 1983. The iron cage revisited: institutional isomorphism
 - and collective rationality in organizational fields. Am. Sociol. Rev. 147–160. Dorst, K., 2006. Design problems and design paradoxes. Des. Issues 22 (3), 4–17.

A. Mallalieu et al.

Dorst, K., 2011. The core of 'design thinking' and its application. Des. Stud. 32 (6), 521-532.

Dorst, K., Cross, N., 2001. Creativity in the design process: co-evolution of

problem-solution. Des. Stud. 22 (5), 425-437

Eder, W.E., 1998. Design modeling-a design science approach (and why does industry not use it?). J. Eng. Des. 9 (4), 355-371.

Eder, W.E., 2009. Why Systematic Design Engineering?. In: International Design Engineering Technical Conferences and Computers and Information in Engineering Conference, Vol. 49057, pp. 513-522.

Eisenhardt, K.M., 1989. Agency theory: an assessment and review. Acad. Manag. Rev. 14 (1), 57–74.

- Eisenhardt, K.M., 1999. Strategy as strategic decision making. MIT Sloan Manag. Rev. 40 (3), 65.
- Eppinger, S., 2011. The fundamental challenge of product design. J. Prod. Innov. Manag. 28 (3), 399-400.

Faludi, J., Hoffenson, S., Kwok, S.Y., Saidani, M., Hallstedt, S.I., Telenko, C., Martinez, V., 2020. A research roadmap for sustainable design methods and tools. Sustainability 12 (19), 8174.

- Gericke, K., Eckert, C., Campean, F., Clarkson, P.J., Flening, E., Isaksson, O., Wilmsen, M., 2020. Supporting designers: moving from method menagerie to method ecosystem. Des. Sci. 6, e21.
- Gericke, K., Adolphy, S., Qureshi, A.J., Blessing, L., Stark, R., 2021. Opening up design methodology. In: The Future of Transdisciplinary Design: Proceedings of the Workshop on The Future of Transdisciplinary Design, University of Luxembourg 2013. Springer International Publishing, pp. 3-14.
- Gericke, K., Eckert, C., Stacey, M., 2022. Elements of a design method-a basis for describing and evaluating design methods. Des. Sci. 8, e29.
- Gmelin, H., Seuring, S., 2014. Achieving sustainable new product development by integrating product life-cycle management capabilities. Int. J. Prod. Econ. 154, 166–177.
- Hackman, J.R., Oldham, G.R., 1976. Motivation through the design of work: test of a theory. Organ. Behav. Hum. Perform. 16 (2), 250-279.
- Hallstedt, S.I., Thompson, A.W., Lindahl, P., 2013. Key elements for implementing a strategic sustainability perspective in the product innovation process. J. Clean. Prod. 51, 277-288.

Hallstedt, S.I., Isaksson, O., Öhrwall Rönnbäck, A., 2020. The need for new product development capabilities from digitalization, sustainability, and servitization trends. Sustainability 12 (23), 10222.

Hallstedt, S.I., Isaksson, O., Nylander, J.W., Andersson, P., Knuts, S., 2023a. Sustainable product development in aeroengine manufacturing: challenges, opportunities and experiences from GKN Aerospace Engine System. Des. Sci. 9, e22.

Hallstedt, S.I., Villamil, C., Lövdahl, J., Nylander, J.W., 2023b. Sustainability fingerprintguiding companies in anticipating the sustainability direction in early design. Sustain. Prod. Consump. 37, 424–442.

- Han, Y., Shevchenko, T., Yannou, B., Ranjbari, M., Shams Esfandabadi, Z., Saidani, M., Li, G., 2023. Exploring how digital technologies enable a circular economy of products. Sustainability 15 (3), 2067.
- Isaksson, O., Keski-Seppälä, S., Eppinger, S.D., 2000. Evaluation of design process
- alternatives using signal flow graphs. J. Eng. Des. 11 (3), 211–224. Isaksson, O., Kossmann, M., Bertoni, M., Eres, H., Monceaux, A., Bertoni, A., Zhang, X., 2013. Value-driven design-a methodology to link expectations to technical requirements in the extended enterprise. In: INCOSE International Symposium, Vol. 23, No. 1, pp. 803-819.
- Isaksson, O., Wynn, D.C., Eckert, C., 2023. Design perspectives, theories, and processes for engineering systems design. In: Handbook of Engineering Systems Design. Springer International Publishing, Cham, pp. 1–47.
- Jagtap, S., Warell, A., Hiort, V., Motte, D., Larsson, A., 2014. Design methods and factors influencing their uptake in product development companies: a review. In: DS 77: Proceedings of the DESIGN 2014 13th International Design Conference.
- Jones, J.C., 1970. Design Methods. John Wiley & Sons.

Julier, G., Kimbell, L., 2019. Keeping the system going: social design and the reproduction of inequalities in neoliberal times. Des. Issues 35 (4), 12-22.

Karlsson, R., Luttropp, C., 2006. EcoDesign: what's happening? An overview of the subject area of EcoDesign and of the papers in this special issue. J. Clean. Prod. 14 (15-16), 1291-1298.

Klotz, L., Weber, E., Johnson, E., Shealy, T., Hernandez, M., Gordon, B., 2018. Beyond rationality in engineering design for sustainability. Nat. Sustain. 1 (5), 225-233.

Kotter, J.P., 1995. Why transformation efforts fail. Harv. Bus. Rev. 73 (2), 59-67. Kotter, J.P., 2012. Leading Change. Harvard Business Press.

https://kingdomwayministries.net/wp-content/uploads/2016/12/Leading Change 2col.pdf.

Kuhn, T.S., 1962/1970. The Structure of Scientific Revolutions. University of Chicago, Chicago.

- Layton Jr., E.T., 1974. Technology as knowledge. Technol. Cult. 31-41.
- Lee, K., 2021. Critique of design thinking in organizations: strongholds and shortcomings of the making paradigm. She Ji 7 (4), 497-515.
- Liedtka, J., 2015. Perspective: linking design thinking with innovation outcomes through cognitive bias reduction. J. Prod. Innov. Manag. 32 (6), 925-938.
- Lindahl, M., 2006. Engineering designers' experience of design for environment methods and tools-requirement definitions from an interview study. J. Clean. Prod. 14 (5), 487-496
- López-Mesa, B., Bylund, N., 2011. A study of the use of concept selection methods from inside a company. Res. Eng. Des. 22, 7-27.

Mallalieu, A., Hajali, T., Isaksson, O., Panarotto, M., 2022. The role of digital infrastructure for the industrialisation of design for additive manufacturing. Proc. Des. Soc. 2, 1401-1410.

- Mallalieu, A., Bonde, J.M., Watz, M., Nylander, J.W., Hallstedt, S.I., Isaksson, O., 2023. Derive and integrate sustainability criteria in design space exploration of additive manufactured components. Proc. Des. Soc. 3, 1197-1206.
- Meyer, J.W., Rowan, B., 1977. Institutionalized organizations: formal structure as myth and ceremony. Am. J. Sociol. 83 (2), 340-363.
- Micheli, P., Perks, H., Beverland, M.B., 2018. Elevating design in the organization. J. Prod. Innov. Manag. 35 (4), 629-651.
- Miles, M.B., Huberman, A.M., Saldaäna, J., 2014. Qualitative Data Analysis: A Methods Sourcebook, Third edition. SAGE Publications, Inc.
- Mitnick, B.M., 1973. Fiduciary rationality and public policy: the theory of agency and some consequences. In: 1973 Annual Meeting of the American Political Science Association, New Orleans, LA. Proceedings of the American Political Science Association.

Mitnick, B.M., 1992. The theory of agency and organizational analysis. In: Ethics and Agency Theory: An Introduction, 1992. Oxford University Press, New York, pp. 75-96.

Mitnick, B.M., 2019. Origin of the Theory of Agency: An Account by One of the Theory's Originators. Available at Social Science Research Network Electronic Journal. https://doi.org/10.2139/ssrn.1020378.

Neri, A., Negri, M., Cagno, E., Franzò, S., Kumar, V., Lampertico, T., Bassani, C.A., 2023. The role of digital technologies in supporting the implementation of circular economy practices by industrial small and medium enterprises. Bus. Strateg. Environ. 32 (7), 4693-4718.

Niiniluoto, I., 1993. The aim and structure of applied research. Erkenntnis 38 (1), 1–21. O'Hare, J., Dekoninck, E., McMahon, C., Turnbull, A., 2010. Adapting innovation tools to

the eco-innovation requirements of industry: case study results. Int. J. Des. Eng. 3 (2), 172-194.

Osterwalder, A., Pigneur, Y., 2010. Business Model Generation: A Handbook for Visionaries, Game Changers, and Challengers, vol. 1. John Wiley & Sons.

Parolin, G., McAloone, T.C., Pigosso, D.C., 2024. How can technology assessment tools support sustainable innovation? A systematic literature review and synthesis. Technovation 129, 102881.

Pieroni, M.P., McAloone, T.C., Pigosso, D.C., 2019. Business model innovation for circular economy and sustainability: a review of approaches. J. Clean. Prod. 215, 198–216.

Pigosso, D.C., Rozenfeld, H., McAloone, T.C., 2013. Ecodesign maturity model: a management framework to support ecodesign implementation into manufacturing companies. J. Clean. Prod. 59, 160-173.

Quella, F., Schmidt, W.P., 2003. Integrating environmental aspects into product design and development the new ISO TR 14062. Int. J. Life Cycle Assess. 8, 113–114.

Ramani, K., Ramanujan, D., Bernstein, W.Z., Zhao, F., Sutherland, J., Handwerker, C., Choi, J., Kim, H., Thurston, D., 2010. Integrated sustainable life cycle design: a review. ASME J. Mech. Des. 132 (9), 091004. September 2010.

Rashid, A., Asif, F.M., Krajnik, P., Nicolescu, C.M., 2013. Resource conservative manufacturing: an essential change in business and technology paradigm for sustainable manufacturing. J. Clean. Prod. 57, 166–177.

- Rittel, H.W., Webber, M.M., 1973. Dilemmas in a general theory of planning. Policy. Sci. 4 (2), 155–169.
- Ross, S.A., 1973. The economic theory of agency: the principal's problem. Am. Econ. Rev. 63 (2), 134-139.
- Rusch, M., Schöggl, J.P., Baumgartner, R.J., 2023. Application of digital technologies for sustainable product management in a circular economy: a review. Bus. Strateg. Environ. 32 (3), 1159-1174.
- Säfsten, K., Gustavsson, M., 2020. Research Methodology: For Engineers and Other Problem-solvers. Studentlitteratur AB.
- Sala, S., Ciuffo, B., Nijkamp, P., 2015. A systemic framework for sustainability assessment. Ecol. Econ. 119, 314-325.
- Samuelson, W., Zeckhauser, R., 1988. Status quo bias in decision making. J. Risk Uncertain. 1, 7-59.
- Schöggl, J.P., Rusch, M., Stumpf, L., Baumgartner, R.J., 2023. Implementation of digital technologies for a circular economy and sustainability management in the manufacturing sector. Sustain. Prod. Consump. 35, 401-420.

Schöggl, J.P., Baumgartner, R.J., O'Reilly, C.J., Bouchouireb, H., Göransson, P., 2024a. Barriers to sustainable and circular product design-a theoretical and empirical

- prioritisation in the European automotive industry. J. Clean. Prod. 434, 140250. Schöggl, J.P., Stumpf, L., Baumgartner, R.J., 2024b. The role of interorganizational collaboration and digital technologies in the implementation of circular economy practices-empirical evidence from manufacturing firms. Bus. Strateg. Environ. 33 (3), 2225–2249.
- Schön, D.A., 1992. Designing as reflective conversation with the materials of a design situation. Knowl.-Based Syst. 5 (1), 3-14.
- Shapira, H., Ketchie, A., Nehe, M., 2017. The integration of design thinking and strategic sustainable development. J. Clean. Prod. 140, 277-287.
- Shapiro, S.P., 2005. Agency theory. Annu. Rev. Sociol. 31, 263-284.
- Simon, H.A., 1969. The Sciences of the Artificial. MIT press.

Simon, H.A., 1979. Rational decision making in business organizations. Am. Econ. Rev. 69 (4), 493–513.

- Sumter, D., Bakker, C., Balkenende, R., 2018. The role of product design in creating circular business models: a case study on the lease and refurbishment of baby strollers. Sustainability 10 (7).
- Sumter, D., de Koning, J., Bakker, C., Balkenende, R., 2020. Circular economy competencies for design. Sustainability 12 (4), 1561.

Thia, C.W., Chai, K.H., Bauly, J., Xin, Y., 2005. An exploratory study of the use of quality tools and techniques in product development. TQM Mag. 17 (5), 406-424.

Thomke, S., Fujimoto, T., 2000. The effect of "front-loading" problem-solving on product development performance. J. Prod. Innov. Manag. 17 (2), 128-142.

A. Mallalieu et al.

Trevisan, A.H., Lobo, A., Guzzo, D., de Vasconcelos Gomes, L.A., Mascarenhas, J., 2023. Barriers to employing digital technologies for a circular economy: a multi-level perspective. J. Environ. Manag. 332, 117437.

Ullman, D.G., 1992. The Mechanical Design Process, Fourth edition. McGraw-Hill, New York.

Ulrich, K.T., Eppinger, S.D., 2016. Product Design and Development. McGraw-hill.

Van Aken, J.E., 2004. Management research based on the paradigm of the design sciences: the quest for field-tested and grounded technological rules. J. Manag. Stud. 41 (2), 219–246.

Verganti, R., Dell'Era, C., Swan, K.S., 2021. Design thinking: critical analysis and future evolution. J. Prod. Innov. Manag. 38 (6), 603–622.

- Vilochani, S., McAloone, T.C., Pigosso, D.C., 2024. Consolidation of management practices for Sustainable Product Development: a systematic literature review. Sustain. Prod. Consump. 45, 115–125.
- Walker, D., Myrick, F., 2006. Grounded theory: an exploration of process and procedure. Qual. Health Res. 16 (4), 547–559.
- Wallace, K., 2011. Transferring design methods into practice. In: The Future of Design Methodology. Springer London, London, pp. 239–248.
- Watz, M., Hallstedt, S.I., 2022. Towards sustainable product development-insights from testing and evaluating a profile model for management of sustainability integration into design requirements. J. Clean. Prod. 346, 131000.

Weber, E.U., 2017. Breaking cognitive barriers to a sustainable future. Nat. Hum. Behav. 1 (1), 0013.

Wohlin, C., 2014. Guidelines for snowballing in systematic literature studies and a replication in software engineering. In: Proceedings of the 18th International Conference on Evaluation and Assessment in Software Engineering, pp. 1–10.